ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dohmen, Thomas J.; Pfann, Gerard A.

Working Paper Worker Separations in a Nonstationary Corporate Environment

IZA Discussion Papers, No. 201

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Dohmen, Thomas J.; Pfann, Gerard A. (2000) : Worker Separations in a Nonstationary Corporate Environment, IZA Discussion Papers, No. 201, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/21045

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 201

Worker Separations in a Nonstationary Corporate Environment

Thomas J.Dohmen Gerard A. Pfann

September 2000

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Worker Separations in a Nonstationary Corporate Environment

Thomas J. Dohmen

Business Investment Research Center, University of Maastricht

Gerard A. Pfann

Business Investment Research Center, University of Maastricht C.E.P.R. and IZA, Bonn

Discussion Paper No. 201 September 2000

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research area *Mobility and Flexibility of Labor Markets*. Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 201 September 2000

ABSTRACT

Worker Separations in a Nonstationary Corporate Environment

This paper investigates differences in worker turnover characteristics between periods of workforce expansion and contraction in a firm. We derive a Cox proportional hazard model from a simple model of job separation based on the expected surpluses from the firm and its workers. We account for non-stationarity in the baseline hazard reversing the role of calendar time and employment duration (Imbens (1994)), and estimate the model using personnel data from a large Dutch aircraft manufacturer over a period from January 1987 until March 14th, 1996, one day before its bankruptcy. Performance evaluations, blue versus white collar employment, and career paths within the firm are found to play an important role explaining differences in turnover behavior. A generally important result is that smooth functional forms of the age effect on worker mobility can produce misleading results, blurring a better understanding of the design of early retirement schemes for corporate reorganizations.

JEL Classification: J63, J26, M12

Keywords: Turnover, personnel economics

Gerard A. Pfann Business Investment Research Center University of Maastricht P.O. Box 616 6200 MD MAASTRICHT The Netherlands Tel.: +31-43-388 3832 Fax: +31-43-388 4856 Email: G.Pfann@KE.UniMaas.NL

1 Introduction

This paper analyzes empirically the firm-specific relationship between separation rates of workers and their characteristics. Particular attention is paid to the question how this relation differs between times of expansion and contraction. The data come from a large firm and lend itself for such an analysis as it spans ten years that can be divided in a period of employment expansion followed by a period of workforce contraction (see Figure 5.1).

Hazard rates for the event of a job separation are estimated. A Cox proportional hazard model is chosen to allow for a flexible baseline hazard, as our main interest is in the prevalent correlations between individual characteristics and separation rates. Our preferred specification is driven by flexibility, since too many restrictions can obscure the true underlying process. Flexibility comes at the price of efficiency loss, though. The size and quality of our dataset, however, enables us to achieve a high degree of precision of estimates.

This is an exploratory study rather than a test of some particular theory on turnover. But the uniqueness of our data set allow us to discuss how existing theories of turnover correspond with the empirical findings. The results suggest important discontinuities of separation hazards with respect to age. This asks for an economic explanation on the one hand, but also has implications for empirical work and human resource management policy within the firm. We spell out a warning that forcing a smooth functional form of the age effect, like a quadratic form, entails a great risk of severe misspecification in the presence of such discontinuities. In fact, in our case, fitting a quadratic age effect produces misleading results.

The paper is organized as follows. Based on theoretical predictions about worker separation in the considered in section 2, section 3 develops a simple model of job separation. Section 4 introduces the econometric model. Section 5 describes the data and covariates used in the empirical application, while section 6 discusses our estimation results. Finally, section 7 concludes.

2 Theoretical Predictions about Worker Separations

Theories on worker turnover have arisen from different strands in economics, often stressing different aspects and proposing different mechanisms that lead to job separations. In general, these theories imply a relation between job separations rates and the joint surplus that the relationship creates on top of the value of alternative options. The size of the joint surplus for a worker and her employer depends on worker characteristics (like education, experience, or age), firm-specific characteristics (such as size, technology, etc.) as well as on external conditions (industry effects, labor market conditions or institutional aspects).

The concept of specific investments is broad and concerns any investment that increases the joint payoff of the relation but is worthless when the relation is not continued. A separation is efficient if the joint payoff associated with continuation of the relation falls below the sum of the value of both parties' alternative option, so that the joint payoff cannot be shared in such a way that the worker and the firm both earn at least their outside option. The separation decision is therefore intrinsically related to the joint surplus.

Inefficient separations can result when the value of the outside option and the value of the relationship depend on random shocks while information about individual payoffs from the relation is asymmetric. Hall and Lazear (1984) show that inefficient quits result when a firm cannot observe the outside option of a worker. After observing the realization of the joint payoff, the firm sets wages to maximize its expected payoff from the relation. Since the firm cannot verify the worker's alternative, it sets a wage that optimally trades off the probability that the employment relation is continued against the payoff it receives given continuation. This offer is strictly smaller than the difference between the joint payoff and the firm's alternative. Consequently workers with an outside offer exceeding the firm's wage offer quit. If information were symmetric, the firm would optimally match their outside offer as long as it earns at least as much as under the alternative; and it would thus prevent workers from quitting. Similarly, if the firm could credibly commit to paying a higher wage than the monopolistic wage in the second period it would do so in return for a money transfer of workers in the first period. Workers are prepared to paying this transfer when being compensated by higher expected earnings in the second period. But since workers cannot verify whether the firm fulfills its promise, they do not accept paying the transfer.

Teulings and Hartog (1998) illustrate how inefficient quits can be prevented if both parties agree on a fixed-wage contract before the true value of the employment relation and the alternative is revealed. Inefficient quits are avoided in situations where the fixed wage exceeds the worker's alternative offer and is higher than the wage under monopolistic wage setting in the Hall and Lazear model. When the fixed wage turns out to be lower than the monopolistic wage, there is scope to adjust the wage upward to the monopolistic wage to preclude inefficient quits. Yet, inefficient layoffs result when the wage to the monopolistic wage. Such renegotiation, would undermine the contract.

Commitment problems affect turnover even under symmetric information when separations are efficient because knowledge about the alternative options ensures that redistribution of the surplus is feasible. Since the cost of a separation depends on the amount of specific investment, the continuation probability increases with the level of specific investment. To achieve an optimal level of investment, it is crucial that the investing party receives the full return to its investment. However, as investment costs are sunk (the investment only pays off if the relation is continued), a hold-up problem arises that results in less than optimal investment levels. MacLeod and Malcomson (1993) show how nominal contracts can alleviate the hold-up problem so that higher investment levels and lower separation rates result. If specific investments are made at all times during the employment relation, investment levels are likely to increase with job tenure which implies a negative relation between tenure and separation rates.

Jovanovic (1979a) derives the structural negative dependence between the separation probability and job tenure as well as experience. While the tenure effect is caused by the growth of firm-specific human capital, the effect of experience follows from the shorter remaining lifetime of the worker over which such investments can be amortized. The model predicts that investments levels tend to be higher, and therefore separation probabilities tend to be lower, the better the quality of the match and the longer the expected remaining duration of the current match. Thus, uncertainty about the duration of the match impacts on separation rates. Pfann (1998) shows how uncertainty about the future creates an option value of waiting when separations are irreversible, thereby offering a closed form solution of Jovanovic's model.

Dustmann and Meghir (1998) propose a mechanism that renders expectations about future surpluses important. They develop a model of wage growth based on learning by doing in which different firms offer different career structures in terms of the rate of human capital accumulation. Separations are expected to occur more frequently early in the career since workers are then more likely to receive a better wage offer than their current, reflecting higher returns to on-the-job search, and at the same time lower forgone returns from job-specific investments. In addition, workers are more likely to separate the longer the period over which they can amortize the investment of job change is.

The idea that careers matter is prominent in the theory of internal labor markets. Baker et al. (1994), for example, find empirical evidence that promotions are serially correlated and associated with wage growth. In such a setting, a worker is willing to make job-specific investments and accept current payment below the alternative option if the job offers sufficient wage growth resulting from sufficiently large future surpluses.

Other predictions stem from models developed in the field of personnel economics.¹ Lazear (1979) argues that seniority pay results from a lifetime incentive scheme where workers are paid less than their productivity early in their career and are rewarded later with pay exceeding their marginal value product. The model relies on the assumption that workers accept an incentive scheme where effort provision and remuneration are distant in time. This creates a severe commitment problem since the firm has an incentive to renegotiate the wage contract or even dismiss workers once it has to start repaying workers on their previous investment. It remains questionable whether this problem can be solved by imposing suitable constraints. If a stable explicit or implicit contract² exists that constraints the layoff decision of the firm, the model nevertheless suggests that separation rates are higher for older workers and workers with longer tenure when the firm is hit by a shock affecting the marginal value product. This is because there are more

¹See Lazear (1995).

 $^{^{2}}$ An implicit contract that rewards workers at the end of the career for effort provided earlier may be credible if for example reneging on the contract would damage the firm's reputation and make it impossible for the firm to hire new workers.

acceptable and cheaper buyout offers to older workers after a negative shock to the marginal value product. A buyout offer is accepted by the worker if it matches the discounted value of the difference between wages under the (implicit) contract and the worker's alternative from the present until retirement. A buyout is profitable for the firm if it costs less than continuation of the employment relation (i.e. the discounted value of the difference between contracted wages and actual productivity of the worker). As the alternative value tends to rise when workers approach retirement age, it is likely that older workers are more willing to accept buyout offers.

3 A Simple Model of Job Separation

We propose a simple model that incorporates many of the above discussed features. We build on the idea that separation rates depend on the joint surplus S. Workers receive a wage W_t in every period t employed. We assume that workers prefer higher wages to lower wages such that utility strictly increases in W. Not working at the firm renders a utility U_A which can be expressed in wage units as A_t . The value of alternative use of time expressed in wage units, A_t , is determined by a number of factors, including the state of the labor market, job offer distribution for individual workers, and the sum of benefits when unemployed or retired (clearly it depends as well on factors that determine eligibility to benefits, e.g. on wealth and prior labor market experience). We also like to think about A_t as including the wage equivalent to the utility loss associated with working at the firm. (Alternatively we could have set the utility loss associated with working to zero.) The utility loss depends undoubtedly on the effort that the worker has to provide on the job, but also on non-labor income if utility is not linear in wages. We can express the instantaneous utility of working in period t as $W_t - A_t$, while the total surplus for the worker in period t, ignoring discounting, is given by

$$S_t^W = W_t - A_t + \sum_{s=t+1}^T E_t (W_s - A_s).$$

This formulation takes account of the fact that not only the surplus created in period t is important, but also expected future surpluses.

The firm pays an employed worker a wage W_t in every period t. The worker produces output V_t in period t. If the worker separates, production is either zero, or the firm has to hire a new worker, so the expected value of output in case of separation, V^A , is determined by the probability of finding a suitable worker times the wage that the worker would demand. Without loss of generality we normalize V^A to equal zero. So the instantaneous value is given by $V_t - W_t$, assuming that the wage paid by the firm equals the wage the worker receives. However, the total surplus for the firm, S^F is given by

$$S_t^F = V_t - W_t + \sum_{s=t+1}^T E_t (V_s - W_s)$$

such that the total expected surplus of the employment relation is given by

$$S_t^E = V_t - A_t + \sum_{s=t+1}^T E_t (V_s - A_s)$$

We assume that firms and workers fix the nominal wage to solve the hold-up problem, that can result from asymmetric information about V and A or from specific investments that are not verifiable to the other party. Workers and firms set wages so as to maximize the total surplus. The share of the expected surplus the firm receives depends on its bargaining power.

The expected surplus can change over time when more information about V and A arrives. An employment relation remains efficient as long as the expected surplus remains positive. The future surplus is stochastic as there are stochastic elements in both the alternative option A_s and V_s . V_s depends on random shocks to productivity, demand or input and output prices. A_s depends on other wage offers that arrive according to some probability distribution. Also, A_s depends on wage growth offered by other firms, which may for example depend in turn on human capital accumulation. With new information arriving constantly, this also implies that the surplus includes an option value of separation, defined as O_t . To illustrate the idea, if there is uncertainty about $\sum_{s=t+1}^{T} E_t(V_s - A_s)$ at time t, the expected loss for the firm is truncated at $V_{t+1} - W_{t+1}$ for the firm since it can layoff a worker in the next period, while the losses for workers are truncated at $W_{t+1} - A_{t+1}$ as they can quit in the next period. Obviously, the option value of a separation is higher, the more uncertain future surpluses and their sizes are.

Next, we assume that a worker separates if the joint expected surplus at time t falls below the cost of separation, C_t , including fixed and variable mobility costs and the option value associated with a separation. That can happen for several reasons. For example, the alternative option might rise when the worker becomes eligible for retirement benefits or when prospective retirement benefits rise, for example because of an additional period of work. Other factors working via the alternative option may be an increase in nonlabor income or the expectation of receiving an even better offer in the near future. These are all cases of efficient separations. But inefficient separations can result when wages cannot be renegotiated. Then, workers optimally quit if their surplus is negative even if the total surplus is positive. If there are mechanisms to renegotiate the wage, the firm could raise the wage, e.g. by promoting the worker if just raising the wage were infeasible, and so prevent an inefficient separation. On the other hand, when the surplus that accrues to the firm becomes negative while the joint surplus is positive, workers may, provided changing the fixed terms of the contract is possible, accept lower wages to prevent being laid off. Such a situation can arise in the case of a downward productivity shock.

The mechanism described above implies that the probability of separation at time t, P_t^S , given that the worker is employed at the firm at t, is a function of the joint expected surplus, i.e.

$$P_t^S = F(S_t^E - C_t) = F(C_t, A_t, V_t; E(A_{t+1}), \dots, E(A_{t+T}); E(V_{t+1}), \dots, E(V_{t+T}))$$

 P^S is the hazard of job separation. The marginal value product V depends on worker characteristics like firm-specific human capital (that is only valuable at the firm and is sunk upon separation) and general human capital, on firm characteristics like capital, production technology, and on external factors including product market demand, competition, factor prices, etc. We like to define firm-specific human capital broadly, e.g. including work relations to colleagues and the like. The value of the alternative usage of time, A, is determined by the probability distribution of job offers, the value of job offers, which in turn depends on worker characteristics and in particular on the rate of human capital accumulation in that job. Furthermore, the alternative value of being employed at the current firm depends on the value of non-labor income, including unemployment benefits or pensions. Finally, the costs of separation depend on fixed and variable mobility costs, on worker characteristics, and on the option value of a separation which is influenced by uncertainty about future surpluses.

4 The Econometric Model

We are interested in the timing of a job separation. Therefore we want to determine the hazard rate of a job separation at time t. Denoting the entry date by t_0 and the exit or failure date as t_1 we define this hazard rate as

$$\lambda = \lambda(t, t_0, X) = \lim_{\varepsilon \to 0} P[t_1 \in [t, t + \varepsilon \mid t_1 \ge t, t_0, X] / \delta$$
(1)

for $t > t_0$. As seen in section 2, this hazard rate depends on individual characteristics X that determine the value of the surplus and the duration $t - t_0$ of the spell at time t. But it also depends on other factors at time t, such as product prices, product demand, other factor prices as well as labor market conditions, which all affect the value of the match. We refer to these factors as the environment at time t.

A popular approach in the literature on duration models is to specify a parametric form of the hazard and estimate its parameters by maximum likelihood techniques. The hazard function as defined in equation (1) may then be specified as

$$\lambda(t, t_0, X) = h(t, t_0, X; \beta) \tag{2}$$

where h is a known function and β is an unknown parameter.

One class of such models, the class of proportional hazard models, assumes that the function h can be factored in a part that depends only on duration $t - t_0$, that is identical for all individuals and usually referred to as the baseline hazard, and a part that depends only on characteristics X. Thus,

$$\lambda(t, t_0, X) = h(t, t_0, X; \beta) = \lambda_0(t - t_0) * f(X; \beta)$$
(3)

where λ_0 is the baseline hazard, β is an unknown parameter and f is a non-negative function.

Functional forms are assumed for both factors, where characteristics usually enter exponentially, i.e. f is the exponential function, while the functional form of the baseline hazard λ_0 is derived from the distribution function of the durations $t_1 - t_0$ in fully parametric models. The hazard function then becomes

$$\lambda(t, t_0, X) = \lambda_0(t - t_0) * \exp(X\beta).$$
(4)

An alternative to the fully parametric models is to leave the functional form of the baseline hazard λ_0 unspecified and estimate it semiparametrically. Such a model (as described in equation (4) above with an unknown form of λ_0) was proposed by Cox (1972, 1975).

However, these models do not allow for a dependence on calendar time, i.e. they implicitly assume a stationary environment. In other words, the hazard rate is assumed the same at any calendar date t for given characteristics X and elapsed duration $t - t_0$. While this assumption may be valid if the period of analysis is sufficiently short, the environment is unlikely to remain stationary if the observation period is long as in our application, and extends over spells from a growing as well as a declining workforce.

Such non-stationarity can in principle be controlled for in the model as described by equation (3) by parameterizing the functional dependence on calendar time t, thus obtaining

$$\lambda(t, t_0, X) = \lambda_0(t - t_0) * f(g(t), X; \beta)$$
(5)

where g(t) is the process that influences the baseline hazard.³ The drawback of this approach is that g must be known exactly. Stated differently, the approach is rather restrictive if the correct functional form of g is unknown or unobservable. An alternative to specifying a particular process that allows for more flexibility is introducing time dummies. The problem with this approach is that time dummies must be specified for particular periods, but with duration data it is not obvious what should constitute such a period.

A more flexible and satisfying approach in our context, where non-stationarity is important, is to reverse the role of calendar time and employment duration in a Cox model, as proposed by Imbens (1994). Anything that affects the separation hazard of employees in the same way on a particular calendar date and is not controlled for by elements in X will be picked up by the baseline hazard $\lambda_0(t)$. Duration dependence has to be parameterized by a function z of duration $t - t_0$. We therefore specify our model as

$$\lambda(t, t_0, X) = \lambda_0(t) * f(z(t - t_0), X; \beta)$$
(6)

³The identification of g comes from a non-degenerate distribution of entry times t_0 .

where we do not have to model calendar time dependence but have to parameterize the duration dependence. A flexible way of parameterizing duration dependence is to introduce dummy variables in z. Hence we specify z as

$$z(t - t_0) = \exp \sum_{i=1}^{I} \alpha_i \omega_i (t - t_0)$$

where $\omega_i(t-t_0) = I[d_i < t-t_0 \le d_{i+1}]$ and I is the indicator function and $d_{i+1} - d_i$ denotes a particular period length. We can still obtain the form of duration dependence from the estimation of the parameters α_i of the function $z(t-t_0)$. A complete specification of our model is hence given by

$$\lambda(t, t_0, X) = \lambda_0(t) * \exp X\beta * \exp \sum_{i=1}^{I} \alpha_i \omega_i (t - t_0).$$
(7)

We thus assume that calendar time effects enter the hazard function multiplicatively. This assumption appears adequate as a first approximation since all workers in our data set can, in principle, be affected by the same set of personnel policies on a given calendar date, since that our data come from one firm only. Moreover, firm-specific effects and labor market conditions affect workers on the same calendar date.

However, firm-specific effects, like personnel policies, and labor market conditions may not affect workers similarly on a given calendar date. We could depart from the assumption that non-stationary forces affect all workers proportionally to the time-independent part of their hazard rate, but we would have to model such non-proportionalities. For example, we can control for non-stationarity in labor market conditions by correcting for differences in unemployment rates of different groups of workers. Clearly, given the host of sources for non-stationarity, such a parametric approach is easier said than done. An alternative - and more promising approach if the main interest is not in the quantitative effects of particular sources of nonstationarity - is stratification. When different baseline hazards are estimated for the different strata we allow for different effects of a common environment on the individual hazard rate. Moreover, in the empirical application, stratification is helpful to assess the sensitivity of estimates to *asymmetric or* non-proportional effects of non-stationary forces.

We can estimate model (7) for different periods to answer whether observed characteristics in the design matrix X affect the hazard of job separation differently in changing environments. We are particularly curious whether such differences exist between times of workforce expansion and workforce contraction. In terms of model (7) this amounts to asking whether the parameters β_j and α_{ij} in

$$\lambda^{j}(t^{j}, t_{0}, X) = \lambda_{0}^{j}(t) * \exp X\beta_{j} * \exp \sum_{i=1}^{I} \alpha_{ij}\omega_{i}(t-t_{0}).$$

differ when the model is estimated for j sub-periods, i.e. if we partition time in J intervals such that $\{T\} = \bigcup_j \{T_j\}$ and $\{T_j\} \cap \{T_i\} = \emptyset$ for all $j \neq i$. We call the earliest date of such a sub-period the starting date of analysis time, t^s . A worker enters the risk pool either at the starting date, if employed at that date, or at the day of entering the firm, if hired later than the starting date. The date of entering the firm, i.e. the beginning of the employment spell is t_0 . Hence $t_0 \leq t^s$ for workers employed at the starting date and $t_0 > t^s$ for workers hired during the observation period. We know the beginning date of an employment spell, t_0 , for all spells, such that we can control for elapsed tenure duration and consequently infer the duration dependence of employment spells. A worker leaves the risk pool on the date of separation. A separation is either a quit, dismissal or retirement.⁴ We censor observations of workers still employed at the end of analysis time, of workers who leave the personnel files but do not loose their job (for example if the employing unit is spun off).

Our data contain complete information from the wage files of the firm. Any change in compensation or working hours is recorded as a new observation. Each observation has a starting date and ending date associated with it. An observation ends when either a change in the wage file is recorded or when the worker separates. For the bulk of the observation such a change is recorded once a year. The average length of an observation is 217.5 days. This allows us to control for changes in observed characteristics, i.e. include time-varying variables in our regression so that model (7) modifies to

$$\lambda(t, t_0, X) = \lambda_0(t) * \exp X(t)\beta * \exp \sum_{i=1}^{I} \alpha_i \omega_i(t - t_0).$$
(8)

One example for a time-varying covariate is the performance evaluation score which is recorded with each wage contract. But we also treat age and

⁴In this paper we do not focus on the determinants of competing risks, but stress the role of the surplus of the employment relation.

tenure as time-varying by calculating the age or tenure at the ending date of an observation. The ending date is also the failure date when a failure occurs and the partial likelihood in the Cox model is calculated at the failure dates. More information on the covariates is provided in the next section.

5 The Data

The data come from a large Dutch company and contain information on all tenured workers who were on payroll between January 1987, when the firm introduced an electronic personnel data system, and March 1996, when the firm was officially declared bankrupt. The company was the Netherlands national aircraft builder N.V. Fokker. The data were made available to us for academic purposes by the company's bankruptcy trustees⁵.

The personnel files contain information on gender, age, tenure duration, marital status, schooling, and vocational training degrees. Moreover, calendar date changes in hours worked, wage changes and the reason for such changes are recorded in the files as well. Reasons for wage changes include promotions, demotions, completion of a degree, and job change within the firm. The bulk of wage changes accrues, however, to such mass mutations as contractual increases reflecting the tenure profile or collectively negotiated wage increases including price compensations.⁶ In addition, we have information about the date of ending the employment relation.

The data are converted into a format that lends itself for duration analysis. For each individual worker we record the beginning and ending date of a wage contract along with observable characteristics during this period. So we have time event data with time invariant and time-varying variables. A contract either ends with a job separation or at the beginning of a new wage contract. We analyze the determinants of separation rates only until the day before the bankruptcy and treat observations of workers still employed on

⁵The report of the bankruptcy trustees is publicly available (see Deterink et. al 1997).

⁶Nominal contracts are characteristic for the Dutch labor market. Yearly wage increases can generally be decomposed into three components (see Teulings and Hartog (1998)): First, contractual experience and tenure profiles result because most collective agreements define wage levels (within a hierarchy of wage scales) that usually award workers with a higher wage for every additional year of tenure. Second, contractual initial increases adjust all wages to changes in aggregate conditions. Third, incidental wage increases relate to any other type fo wage increase like promotions.

the day of the bankruptcy, when more than 3000 jobs are killed, as censored. Apart from the technical problem of many ties for this failure date, results would likely be blurred and misleading because the hazard rate of loosing ones job is by definition infinite for workers falling under the bankruptcy. To obtain more reliable estimates, the day of bankruptcy is excluded from the study.⁷

The time period spanned by the data can be divided into a period of workforce expansion and a period of employment reductions. The employment dynamics are presented in Figure 5.1, which shows the number of individuals in our data set employed on the 14th of each month. Employment increased from about 10000 full-time equivalent workers in 1987 to reach a peak of about 12500 full-time equivalent workers in 1991 and subsequently fell to less than 6000 workers in 1996 after a number of reorganizations with mass layoffs. We analyze the separation hazards until the day before the bankruptcy, i.e. just before the final sharp decline of the workforce in Figure 5.1.

On the day the electronic personnel system was put into use (01/01/1987), the company's organizational structure was made up of a holding corporation (N.V. Koninklijke Nederlandse Vliegtuigenfabriek Fokker), an administration company (Fokker Administration B.V.), a production company (Fokker Aircraft B.V.), with plants in three different locations, as well as five other subsidiaries spread over different locations (see Appendix). Most of the company's organizational structure remained unchanged until the day of the bankruptcy.

We define eight dummy variables for each of the different subsidiaries or locations. Clearly, not all subsidiaries were hit by the company's demise equally hard. Therefore we control for the location where each worker is physically employed (this may vary through time). The company's core business, aircraft construction, is done by Fokker Aircraft B.V. at three different locations, which we call location 0, 2 and 6 as is explained in more detail in the Appendix. About 80% of the entire workforce is employed in one of these three locations on March 1st 1991 (see Figure 5.2a). Employees of Fokker Administration B.V. also work in location 0. These subsidiaries and the holding eventually go bankrupt.⁸ Location 1 only exists until 1990 when it is

⁷This procedure is not least justified by the fact that the bankruptcy came as a surprise to many.

 $^{^{8}}$ It should be noted that some parts of the firm survived or were spun off, so that not

integrated with the main activity at location 0. Location 3, 4, 5 and 7 coincide with the remaining four subsidiaries. It should be noted that subsidiary 3 is spun off in December 1995 when workers employed at this subsidiary consequently leave the personnel files of the firm. We treat observations on those workers as censored as they kept their job under the new ownership.

In our empirical analysis, we include also information on education age, tenure, age, education, gender, marital status, part-time work, and on performance evaluations. Furthermore, we include a number of measures that track a worker's career path inside the firm. Information on wage scales of the firm allows us to distinguish between white-collar and blue-collar workers. All these characteristics - variables are defined in the Appendix - are expected to affect the surplus, via an effect on productivity or via an impact on the outside option. We avoid much functional form restrictions on the effect of such variables by defining categorical variables.

Tenure duration is divided into 16 categories. These are obtained by calculating tenure in days at the end of an observation and then transforming this variable by the function $\omega_i(t-t_0) = I[d_i < t - t_0 \leq d_{i+1}], i = 1, ..., 16$, where I is the indicator function and d_i denotes a particular cut off date. We allow the length of the period to increase with tenure duration as the distribution of tenure at the firm is skewed (see Figure 5.2b).⁹ Age categories are defined similarly with an interval length of one year.

Figures 5.2a - f provide information on the composition of the workforce on March 1st 1991, when the first reorganization was announced (cf. Deterink et al., 1997). Figure 5.2b shows the tenure distribution. About 11.8% of the workforce has less than one year of tenure, and 43.2% of the workforce has been hired in the previous 5 years. Almost 14.4% of the workforce has been more than 20 years. Median tenure is 6.16 years, while the mean tenure is 9.66 years. Figure 5.2c shows the age distribution. Half of the workforce is younger than 33 years of age; 11.1% are older than 50 and 12.2% are younger than 24. Less than one percent of the workforce is older than 60.

We include nine categorical dummy variables for the highest educational degree obtained¹⁰. Most workers employed on March 1st 1991 hold, as their

all workers automatically lost their job during the firm's downsizing period.

⁹For details on the interval lengths see the Appendix.

¹⁰Since information on the highest educational degree is missing for almost a quarter of the employees, we define an extra category for those workers in order not to lose valuable

highest completed degree, a vocational degree as Figure 5.2d illustrates. A quarter of the workforce has completed an apprenticeship, 17.9% completed lower vocational schooling, 20.3% intermediate and 13.1% higher vocational schooling.

Performance evaluation scores are denoted by 6 categorical dummy variables. They show little variation and are skewed towards the high end (Figure 5.2e). More than 89% of workers employed on March 1st 1991 are judged to be good or very good (59.0% and 30.2% in categories 3 and 4 respectively); more than 9.1% received even higher scores, 1.6% perform satisfactorily while the performance of only 0.1% is unsatisfactory.

We account for the 18 different wage scales that exist at the firm.¹¹ Workers in the 10 lower wage scales (scales 2-11) are referred to as "blue-collar workers" and are bound by collective wage bargaining agreements. Workers in the higher scales (12-19) are referred to as "white-collar workers". Their wages are not bound by collective wage bargaining agreements. Figure 5.2f shows that most workers (73.55%) are in the lower ten wage scales. The proportion of workers in each of these wage scales increases initially from less than a percent in scales 2 and 3 to peak at more than 15% in scale 8. The number of workers in the next three higher scales falls consecutively. The remaining 26.45% of the workers are not bound by collective agreements. Figure 5.2f reveals the pyramidal composition in these higher wage scales with more than 60% of the higher-wage workers in scales 12-14 and less than one twelve % in scales 17-19. As workers in the lower scales (2-11) are predominately employed in production, the clear cut in the firm's low and high salary scales coincides with the division of the workforce into blue-collar and white-collar workers. White-collar workers represent a large share (more than 70%) of the workforce at location 3. Blue-collar workers represent a large share of the workforce at location 2, 4, 6 (more than 80%) and location 5 (about 70%). A considerable share of white-collar employees (more than 30%) works at location 0.

We hold the first observed wage scale of a worker constant¹² and include

information. Information on education is missing for workers in all wage classes.

¹¹Note that wage levels exist within each scale which can be interpreted as contracted tenure profiles as workers generally move to a higher level automatically after additional years of employment at the firm.

¹²This is either the wage scale observed on January 1st, 1987, or the entering wage scale of a worker who joins the firm after that date. It should also be noted that scale 12 is a

dummy variables that trace transitions in wage scales. By March 1991, less than 0.6% have moved down the wage ladder, 31.4% have moved up one scale, 17.8% two scales, and 8.0% three or more scales. We use additional information from the wage data files of the firm to follow the path of formal promotions, completion of a higher educational degree, and wage growth associated with such events. Less than 19% of the workers employed on March 1st 1991 received a new contract because of a promotion during the period starting on January 1st 1987 (16.62% were promoted once, 2.24%twice and 0.3% received more than 2 times a new wage contract because of a formal promotion). The cumulated percentage wage growth associated with such new contracts ranges from less than 1% to more than 50% (while it is less than 25% for 99.5% of those who are ever promoted). Contracts were changed for only 2.86% of workforce because of a degree completion. Cumulated wage increases associated with such contract changes range from about 1% to almost 10%. A further dummy variable conveys information on a prior completed employment spell at the firm. 165 workers (less than 1.3%of the workforce employed in 1991) have such prior experience.

Figures 5.3a-f and 5.4a-f show the distribution of characteristics of workers who separate during the periods 1987-1991 and 1991-1996 respectively. Figures 5.3a and 5.4a reveal that the share of separations from locations 2 and 6 is larger in the second period while a smaller share separates from location 0. Yet in both periods most workers were employed at location 0 when they separate, which is not surprising as the largest fraction of the workforce is employed there (see Figure 5.2a). Figure 5.3b illustrates that most separations occur at short tenure during the period of workforce expansion: 23.6% of separating workers have been with the firm for less than a year; more than half of the separations take place during the first four years and about three quarters during the first ten years of the employment relation. While a small proportion of the separating workers has been employed at the firm for 15 to 30 years (6.16% only), separations become more frequent again after around 35 years of tenure. The tenure distribution among separations is similar in the period of downsizing: Separations occur most frequently at short tenure, least at intermediate tenure duration and again more frequently after tenure duration of more than 30 years (see Figure 5.4b). However, a considerably smaller proportion has less than a year of tenure (only 6.8%

particular one, since most white-collar workers who enter the firm start in that scale.

compared to 23.6%) - which might reflect lower hiring rates in the period of downsizing and hence a smaller pool of workers with less than a year of tenure - while a larger proportion has 4-10 years of tenure (33.5% compared to 22.7%). Therefore, average tenure of separating workers is longer in the period of downsizing (9.4 years as compared to 8.6 years).

Figures 5.3c and 5.4c show the age distribution of separating workers in the two periods. In both periods, the distribution is not unimodal. The first peak is at the age of 25 years in both periods (at 6.2% and 5.9% of the separating workforce in the first and second period respectively). Most workers who separate are 60 years old in the first period (14.0%). In the second period the peak at age 60 is less marked. Instead, a peak in the age distribution occurs earlier at 55 and 15.8% of those who separate are aged 55-60. Given that a separating worker is older than 50, he is expected to be younger in the second period, while the opposite holds for workers younger than 50. The net effect is that separating workers are on average somewhat younger in the first period, 34.8 years in the first compared to 35.5 years in the second period.

Comparing the educational achievements of the separating workers in the two period, it becomes apparent that a larger proportion of workers holds lower degrees in the second period. In addition, relatively fewer workers hold a vocational degree in the first period (see Figures 5.3d and 5.4d).

A considerable proportion of separating workers has an above average performance evaluation score in both period. However, the average performance score of separating workers is always lower (3.36 in the first and 3.39 in the second period) than that of the workforce employed on March 1st 1991 (3.48) (see Figures 5.3e and 5.4e).

Figures 5.3f and 5.4f indicate that workers separating in the second period had on average further advanced careers in the firm. The largest proportion blue-collar workers is in wage scale 5 at the date of separation in the first period, but in wage scale 8 in the second period. This is similar for whitecollar workers: Relatively more separating workers are in the firm's lower wage scales in the first period.

6 Estimation Results

We estimate different specifications of model (7) by Cox partial likelihood regressions where the Breslow method is used to deal with ties. We start with a basic specification which includes all variables described in the previous section. This specification is estimated for the period from January 1st 1987 until March 14th 1996 and for two sub-periods: from January 1st 1987 until March 1st 1991, the period of workforce expansion, and from March 1st 1991 until March 14th 1996, the period of downsizing. The coefficient estimates are in Tables 6.1a-f. Estimating the model for different time periods allows coefficients as well as the baseline hazard to differ. We compare the baseline hazard from the model estimated for the sub-periods to get an impression how changes in separation rates in the different regimes decompose into an effect that is the same for everybody in the sample (and hence reflected in the baseline hazard) and group specific effects (reflected in changes of coefficient estimates).

Furthermore, we restrict the parameter vector β in different ways and allow for different forms of stratification to check the robustness of our results. Allowing the baseline hazard to differ for different strata while restricting the remaining coefficients to be identical across strata also allows us to assess the proportionality assumption and the sensitivity of other estimates to the validity of this assumption. Finally, we report the parameters estimated for the sub-samples of blue-collar and white-collar workers in Tables 6.2a -f and 6.3a-f respectively.

The picture that emerges is briefly sketched as follows. The shape of the baseline hazard differs in periods of workforce expansion and downsizing, being rather flat and comparably small in the period of employment growth but erratic and much higher in the period of downsizing. The shape of the baseline hazard remains similar after splitting the period, accentuating that the risk of job separation is generally higher in the second period (compare panels a) and b) of Figure 6.1)

A comparison of parameters estimated separately for the high and low turnover regime indicates that relative separation rates differ in the two regimes (compare Table 6.1 columns 2 and 3). Some parameter estimates even change sign between the two regimes. For example, women face higher separation hazards in the low turnover regime, while they are less likely to separate than their male colleagues in the period of downsizing.¹³. Whitecollar workers have higher turnover rates in the first period and lower ones in the second period than their blue-collar colleagues. The impact of other characteristics also changes.

Third, the results from imposing different restrictions on the vector of observed characteristics show that coefficient estimates are generally robust in the sense that the exclusion of variables does not have a significant effect on other estimated coefficients. For example, leaving out controls for a worker's career does not significantly change the size of coefficients for marital status, gender, tenure, age, education, location dummies and controls for wage scales. Leaving out location dummies, or controls for gender and part-time workers does not change coefficient estimates of other variables either. The fit, however, worsens significantly when such control variables are left out.

Yet, not controlling for wage scales has an impact on estimated coefficients for education levels and measures of career attainment in the period of restructuring. This reflects the fact that high-wage workers face a much lower increase in the hazard of job separation than workers in lower wage scales, while schooling and income class are highly correlated. The impact of education on separation rates can therefore be decomposed into two effects: the effect of education given the job level and the effect of education on the probability of attaining a particular job level.

Coefficient estimates are also robust when the baseline hazard is allowed to differ across strata for gender, part-time workers, locations or wage scales. In addition, the baseline hazards for men and women, full-time and part-time workers, and workers employed at different locations are similar, implying that the proportionality assumption is valid. This is encouraging if our aim is to quantify the effect of particular characteristics on separation rates. We can therefore refer to coefficient estimates from our basic specification when comparing relative separation rates for different groups and do not report the results of all stratified models. However, as the baseline hazard differs for blue- and white-collar workers, we discuss the differences in the baseline hazards obtained from this stratification in more detail. These results allow us also to be more precise with respect to the differences in the exact timing of events.

¹³This might reflect that many women work as secretaries. This job is complementary to many white-collar jobs, which are cut relatively less in the period of restructuring.

6.1 Baseline Hazards

Entire Sample

Figure 6.1 (panel a) shows the baseline hazard obtained from the basic specification estimated for the entire period.¹⁴ It reveals large differences in the baseline hazard between the period of workforce expansion and the period of downsizing. These differences of the hazard rate at different calendar dates highlight the importance of non-stationarity. Controlling for such non-stationarity is crucial. The figure shows that the hazard rate for a job separation is rather flat in the period of expansion until 1991, while it is much more erratic and considerable higher thereafter, especially from 1993 onwards. The fact that the baseline hazard does not rise sharply until 1993 while the workforce shrinks since 1991 (see Figure 3.1) suggests that the workforce reductions are accomplished by either lower hiring at a given separation rate or are born by particular groups of the workforce. Such group-specific differences in hazard rates show up in the estimated coefficients of the control variables that characterize such groups, as for example age. This squares nicely with our prior knowledge that workforce reductions were mainly accomplished by natural turnover such as retirement as well as early retirement prior to 1993 (see Deterink et al. 1997).

White-Collar vs Blue-Collar Workers

Figure 6.2, 6.3 and 6.4 provide information on the question whether bluecollar workers and white-collar workers are affected in similar ways by changes in the environment. The baseline hazards in Figure 6.2 are obtained from a stratified model that allows only the baseline hazard to differ for the strata of blue- and white-collar workers and is otherwise identical to the model underlying Figures 6.1 as it restricts the parameter estimates to be the same for both groups of workers. Figures 6.3 and 6.4 plot the baseline hazards that are obtained after splitting the sample into blue-collar and white-collar workers, such that the underlying models allow coefficients and baseline hazards to differ across strata and periods. The corresponding parameter estimates are in Table 6.2.

All estimated baseline hazards reveal that white-collar workers face a

 $^{^{14}}$ The corresponding parameter estimates are in Table 6.1, column 1. Panel b) of Figure 6.1 plots the baseline hazards obtained after splitting the period; and the corresponding parameter estimates are in Table 6.1, columns 2 and 3.

lower separation hazard in the period of downsizing while they are more likely to separate in the period of workforce expansion. Although, the baseline hazard rate peaks in 1993, 1994 and 1995 for both types of workers, these peaks are much larger for blue-collar workers, especially in the earlier years 1993 and 1994. This clearly suggests that downsizing is asymmetric in this period, which may result from differences in adjustment costs and a perception that the crisis is temporary (cf. Hamermesh and Pfann, 1996). Similar asymmetries are also found within the group of blue-collar workers. If we stratify according to wage scales, we find that workers in lower wage scales face higher hazard rates earlier in the period of downsizing.¹⁵

Given this result and the fact that blue-collar and white-collar workers are employed in different proportions at the different subsidiaries, it is not surprising that job separation rates differ across subsidiaries. Hazard rates increase markedly in 1993 at production sites, paralleling the increase in the baseline hazard for blue-collar workers in Figure 6.3. The hazard rate increases most at production sites of the core activity (locations 0, 2 and 6) in later years. This becomes apparent when stratifying with respect to locations. It should also be noted that the hazard rate is markedly higher at location 6 than elsewhere in 1995 when there were plans of closing that site.¹⁶

6.2 Duration Dependence of Job Tenure

The duration dependence of job tenure can be inferred from the estimated coefficients α from model(7). Tables 6.1a - 6.1f report the coefficients estimated for the entire period (column 1) and two different sub-periods (columns 2 and 3). Tables 6.2a - 6.2f and 6.3a - 6.3f present the estimated coefficients for the strata of blue-collar and white-collar workers respectively. Estimates for the entire period are again in column 1 while the estimates for the two subperiods are in columns 2 and 3. All coefficients in a given column of the different panels of the tables are estimated jointly, but are presented separately here for illustrative reasons. Column 4 gives in each case the p-value for the test that the coefficient in column 3 equals the point estimate of the coefficient in column 2. This gives an indication whether coefficients are

 $^{^{15}\}mathrm{We}$ do not plot these baseline hazards in this paper.

 $^{^{16}}$ See Deterink et al. (1996) for more details on such plans.

differ in the two sub-periods.

The coefficients correspond to the elements of the coefficient vectors α and β in model (7), and measure therefore the effect on the log hazard rather than hazard ratios. The coefficients ought to be interpreted as follows: A unit change in a covariate x changes the log hazard by the amount of β_x , where β_x is the estimated coefficient of covariate x, or in other words, a unit change in x changes the hazard by approximately $\beta * 100\%$ for values of β close to zero. Thus, we can interpret coefficients of categorical variables as semi-elasticities.

The estimated coefficients for the tenure categories in Tables 6.1a, 6.2a and 6.3a indicate that turnover rates are especially high in the first three months of an employment relation. Although the risk of job separation falls considerably after this initial period, it remain relatively high during the first year. Subsequently, the hazard of job separation falls gradually throughout the first 15 years of tenure with most of the decline taking place in the first ten years of an employment relation. After about 15 years of job tenure, the hazard rate is rather low and remains flat only to increase again at very long tenure duration, a feature that can be ascribed to (mandatory) retirement. This broad pattern emerges in all specifications of the model.

However, there are important differences in the two sub-periods. While, in the period of expansion, the hazard rate of job separation is 29 times higher for workers with less than 3 months of job tenure than for workers in the reference group who have 15-20 years of job tenure, it is only about 22 times higher during the period of downsizing. However, the hazard rate does not fall as much during the first five years of tenure in the period of downsizing as it does in the period of expansion. The difference is considerable and the p-values in column 4 of Table 6.1a indicate that the separation risk from the 4th month until the 5th year of the employment relation is significantly higher during downsizing. The point estimates suggest that, in the period of downsizing, the hazard rate is on average about 6.5 times higher for workers with 4-12 months of tenure than for workers in the reference group, while it is on average only about 4.5 times higher in the period of expansion.

While the hazard rate relative to workers in the reference group falls considerably after the first year of tenure during the period of expansion, it remains very high throughout the first three years of the employment relation during the period of downsizing; in the 5th year of the employment relation the separation hazard is still more than 3 times higher during the period of downsizing and roughly 2.5 times higher during expansion than for workers in the reference group. Relative hazard rates remain higher for workers with more than 5 and less than 15 years of tenure during the period of downsizing, yet the difference is less marked. Thereafter, relative rates are rather similar. Only after 35 years of tenure rise relative hazard rates more sharply in the period of downsizing. While workers with 35-40 years of tenure face not a significantly higher hazard rate than their colleagues in the reference group with 15-20 years of tenure face a hazard rate that is about twice as high, workers with 35-40 years of tenure face a hazard rate that is about twice as high, workers with 35-40 years of tenure have a significantly higher hazard rate in the period of downsizing when those with more than 40 years of tenure face even 3 times higher hazard rates than these in the reference group. Again, the p-values in column 4 indicate that these differences are significant.

A glance at Tables 6.2a and 6.3a reveals that these results are largely driven by changes in hazard rates for blue-collar workers. The p-values in column 4 of Table 6.3a show that relative hazard rates do not differ significantly in the two sub-periods for white-collar workers with less than 25 years of tenure. Relative hazard rates for white-collar workers differ only at long tenure. For blue-collar workers, the patterns is as described above.

High turnover rates early in a relationship are often ascribed to the fact that the lack of quality of the match is revealed after a short period of work.¹⁷ If the match product turns out to be low, a separation results. The negative duration dependence can also be explained by human capital accumulation.¹⁸ We would expect separation rates to fall with tenure as long as firm-specific knowledge is accumulated. The rate of specific human capital accumulation as well as the rate at which returns from such capital is decreasing would affect the rate of decline of job separation rates with tenure. In principle, any model with match-specific investment would generate the same kind of duration dependence, as long as the joint surplus increases while the alternative option of workers and firms remains unchanged.

A life-cycle incentives model, as for instance proposed by Lazear (1979), also generates a negative duration dependence. In Lazear's model, separation rates should fall with tenure until the worker gets repaid for the effort

¹⁷See for example Jovanovic (1979b).

 $^{^{18}}$ Jovanovic (1979a) focuses on the relationship between firm-specific human capital and turnover rates.

he provided early in his career. Then it may rise if buy-outs are feasible. In absence of shocks that make such buy-outs optimal for both parties we should expect the hazard rate to remain flat until retirement. When shocks are large, separation probabilities should be higher the more feasible a buyout opportunity becomes. This is more likely the shorter the period until mandatory retirement and hence the longer tenure (at given age). This is roughly what the hazard rates for the period of downsizing suggest.

The increase in the hazard rate at long tenure suggests that (early) retirement occurs more frequently in the period of downsizing. The higher hazard at short tenure, especially during the first 5 years, indicates that those who were hired in the previous expansion are hit more severely in the period of downsizing than workers who had joined the firm before 1987. These results illustrate that different tenure groups are affected differently. This points at a dependence relation between a worker's past career path within the firm and turnover (costs).

The result that relative turnover rates rise for workers with shorter tenure in periods of downsizing is also consistent with the LIFO-layoff model of Kuhn (1988) and Kuhn and Roberts (1989). Since unions impose the LIFO-layoff rule in these models to extract rents from the firm, we expect that the relative increase of the separation rate for workers with short tenure in periods of downsizing is stronger for blue-collar workers, who are covered by collective bargaining agreements, than for white-collar workers who are not bound by such agreements. Indeed, a comparison of the differences between tenure coefficients for blue-collar workers in the two different periods (columns 2 and 3 of Table 6.2) and differences for white-collar workers (columns 5 and 6) yields support for this view. While the tenure effect during the first 25 years of tenure is similar for white-collar workers in both periods, blue-collar workers with short tenure face significantly higher hazards of job separation in the declining phase than in the growing phase relative to their colleagues in the reference group. This effect is especially strong during the first five years of tenure.

6.3 Age Effects

The effect of age on separation rates produces novel results. The effects are not smooth. Figures 6.5a, 6.5b and 6.5c, which plot the coefficient estimates for the age categories from the models reported in Tables 6.1b, 6.2b and

6.3b, illuminates that discontinuieties are especially important for workers older than 50 years. The hazard rate for blue-collar workers (Figure 6.5b and Table 6.5b) falls only gradually with age during the period of expansion for those aged 25 to 50 and the estimated hazard rate for most age categories is not significantly different from the hazard rate of workers in the reference group. In addition, the p-values in column 4 of Table 6.2b for the test that the estimated coefficient in the period of downsizing is equal to the point estimate for the corresponding coefficient in the period of expansion indicate that age effects are similar for workers aged 25-50 during downsizing. Then, the hazard rate exhibits even less variation for workers younger than 45.

But the hazard rate drops markedly in both periods for blue-collar workers in their early fifties. This drop is more severe in the period of downsizing when the dramatic decline of the hazard rate sets in at a somewhat younger age. Hazard rates are lowest for workers aged 53 (90% lower than that of workers in the reference category) in the period of expansion and for workers aged 52 (92% lower) in the period of downsizing (Table 6.1). After the respective minima are reached, the hazard rate rises steeply thereafter in both periods. During the period of expansion, the hazard rate of those aged 54 to 59 is not significantly different from the hazard rate of workers in the reference group aged 35. This is different during downsizing when the hazard jumps up for workers aged 55 and remains high thereafter. The p-values in column 4 indicates that this difference is significant for workers aged 55 to 58. The hazard rate of workers aged 55 to 58 is on average about 9 times as high as the rate for workers in the reference group.

At the age of 60, the hazard rate jumps up markedly again. Then, the hazard rate is more than 30 times higher than that of workers aged 35 in both periods. The hazard rate falls somewhat thereafter, but remains high. This effect is driven by retirement and the result suggests that workers are most likely to retire at the age of 60 in the period of expansion. The pattern for white-collar workers older than 50 is similar to that described for blue-collar workers as Figure 6.5c and Table 6.3b reveal.

These results reflect the consequences of early retirement schemes, especially in the period of downsizing. The age effect may be driven by changes of the alternative option for workers older than 50 years. This assumption is reasonable since discrete jumps in productivity related to age are unlikely. Such an effect can result from early retirement options at a certain age. The option to retire early at the age of 55 is not uncommon in the Netherlands. Such a pension scheme may explain the discrete jump in exit rates at age 55. Moreover, early retirement also seems to have an option value. This option value of waiting for the possibility to retire early seems to be highest for workers aged 50-54 and their separation probabilities are consequently lowest.¹⁹ This result has important personnel policy implications for the design of early retirement plans.

Age effects for workers younger than 50 are more in line with findings of existing studies. Table 6.1b shows that workers aged 20 to 34 face higher hazards than workers in the reference group aged 35. This effect is stronger and significant in the period of corporate growth when hazard rates for those younger workers are on average about 60% higher than for workers in the reference group, but the effect is insignificant in the period of decline. Relative hazard rates for workers in their late thirties and early forties are smaller but not significantly so. Only in the period of downsizing do workers in their late forties face significantly lower separation rates than workers in the reference group.

More differentiated results surface when the model is estimated separately for blue- and white-collar workers (see Tables 6.2b and 6.3b). The estimates indicate that hazard rates decline stronger with age for young white-collar workers, and hazard rates relative to those of the respective reference group are higher for young white-collar workers than for young blue-collar workers in both periods. Table 6.3b shows in addition that young white-collar workers face especially higher separation rates in the period of corporate growth and the hazard of job separation then declines more with age for young whitecollar workers. Separation hazards of workers younger than 30 are then 2.3-7.3 times higher than for white-collar workers aged 35.

White-collar workers have usually completed higher education and acquire relatively less skills on the job. They probably make fewer firm-specific investments than their blue-collar counterparts. Consequently, the surplus of the current match over the outside option tends to be smaller for white-collar workers and job-shopping is therefore a more common phenomenon early in the career of a white-collar worker.

It is interesting to note that blue-collar workers who are younger than 20

¹⁹Stock and Wise (1990) develop a retirement model which takes into account the option value of continuing work and thereby keeping the option to retire on better terms alive. Their model fits firm retirement data, marked by sharp discontinuous jumps in retirement rates at specificic ages, well.

years face lower separation rates in the period of workforce expansion. These workers are typically acquiring job-specific skills and often complete some vocational schooling (they are frequently apprentices). The results suggest that they become more likely to separate when the prospects of their current employer worsen, possibly because the horizon over which they can amortize their investments becomes more uncertain.

Finally, the hazard rate falls when workers are in their forties (see Table 6.1b). The hazard rate is on average about 33% lower for workers in their late forties than for workers in the reference category. Again, this effect is somewhat stronger for white-collar workers (see Table 6.2b). So we can summarize that young workers are less likely to stay with the same firm and separation rates fall for some time with age to rise again at retirement age. Existing studies often fit a quadratic age effect to capture such a pattern. However, this approach is flawed, because it ignores the discontinuities described above. Typically these studies estimate the minimum separation rates to occur at age 35-40. In fact, we have simulated this approach by also fitting a quadratic age effect in our data and found the minimum to be at age 36.8 for the model in the first column of Table 6.1b. Such a result would be consistent with lifetime incentive models because we should expect the firm's liability for the worker to be highest towards the middle of his working career. In addition, the chance that a profitable buyout is possible when productivity shocks hit is small as the claims of the worker would be large.

However, with the effects as estimated above it is not entirely obvious why it is optimal for firms that workers older than 55 separate more frequently in periods of downsizing while workers who are slightly younger stay. Early retirement schemes have strong spill-over effect on slightly younger workers that should be taken into account. Downsizing firms may want to offer less generous early retirement schemes to a broader group in worker to retain productive workers.

6.4 Effects of education

Education has both a direct impact on turnover rates and an indirect one via the assignment to a higher job level. Higher education raises the probability of being in higher wage scales of the firm. Workers in higher wage scales tend to have higher separation rates in the low-turnover regime but lower separation rates in the high-turnover regime (compare columns 2 and 3 of Table 6.1e and note also that coefficients for low scales (4-9) and high scales (15-19) differ significantly across periods as the p-values in column 4 of Table 6.1e indicate). Assuming that the wage scales proxy for job complexity, we find the effect of education given job complexity when holding the wage scales constant. This is the direct effect of education.

Controlling for wage scales, we find that vocationally educated workers have lower separation rates than those with a general education. Workers holding only a general schooling degree (i.e. basic education (lo), intermediate general education (mavo), higher general schooling (havo) or general schooling that entitles to take university courses (vwo)) are more likely to separate than those who complete a vocational schooling degree at the level for which the general educational degree qualifies.²⁰ Furthermore, given wage scales, the effect of education is u-shaped, meaning that separation rates are lowest among workers with medium level education. While the impact of the educational level on the separation hazard does not change significantly for white-collar workers in the period of corporate decline (see coefficient estimates in columns 2 and 3 of Table 6.3c and p-values in colums 4 of the same Table), higher educated blue-collar workers become relatively less likely to separate during downsizing (see Table 6.3c).

Workers with completed apprenticeship face the lowest hazard of job separation in both periods. Apprenticeships are frequently completed at the firm. More than 57% of the workers, who join the firm after 1987 and obtain a higher degree during the employment spell, complete an apprenticeship, 18% complete intermediate vocational schooling, 8% higher vocational schooling, almost 6% obtain a college or university degree and roughly 10% achieve a higher general qualification. Since vocational schooling offered by the firm is likely to teach firm-specific skills and knowledge, our results suggest that separation rates are lower the more firm-specific a worker's human capital is. This is consistent with the prediction from human capital models.

The total effect of education on the job separation hazard is estimated when we do not control for wage scales. Since the separation rates of workers in different wage scales are affected non-proportionally by non-stationary forces, with workers in lower wage scales becoming relatively more likely to

²⁰For example, intermediate general education (mavo) is a prerequisite for intermediate vocational schooling (mbo) and workers with an intermediate vocational schooling degree have lower separation rates than those with an intermediate general schooling degree only.

separate in the period of downsizing, it is important to distinguish between the high- and low-turnover regime. In the low-turnover regime, when separation rates are rather similar across wage scales, the effect of education is barely affected by leaving out controls for wage scales. The u-shaped effect becomes somewhat more pronounced and vocationally trained workers are less likely to separate. However, the u-shaped effect disappears in the second period, when hazard rates fall with the level of education. This result is largely driven by the indirect effect of education, i.e. the fact that workers in higher wage scales face significantly lower hazard rates. The hazard rate of a worker with a university degree is more than 25% lower than that of a worker who has completed an apprenticeship, the group that was least likely to separate in the first period. Vocationally trained workers still have lower hazard rates than their colleagues who hold a general degree that qualifies for the vocational training. Workers with vocational training at the highest level are least likely to separate in the period of downsizing.

Thus, in periods of turmoil vocationally trained workers and workers in more complex jobs face lower separation rates. Since higher educated workers have a higher probability of being in a more complex job, higher educated workers are less affected by non-stationary forces that raise the hazard of job separation.

6.5 The Effect of Performance Evaluation and Career Paths

The effect of the performance evaluation measures is significant and strong throughout (see Tables 6.1d, 6.2d and 6.3d). Workers with very low scores are much more likely to separate than workers in the reference group who were judged to perform well (see the Appendix for definition of variables). Overall, workers with the lowest score are 6.3 times more likely to separate than workers in the reference group (Table 6.1d, column 1); this effect is stronger for white-collar workers than for blue-collar workers in both subperiods (compare Tables 6.2d and 6.3d). In addition, the effect of having the worst evaluation score compared to having the reference group's evaluation score never changes significantly across periods (see columns 4 of Tables 6.1d, 6.2d and 6.3d).

This is different for blue-collar workers with evaluations better than those

of the reference category. These workers become significantly less likely to separate during downsizing. In this period of corporate decline they have 30.4% (evaluation score of 4) to 47.4% (evaluation score of 6) lower hazards than workers in the reference group, whereas only workers with an evaluation score of 4 are significantly less likely to separate than workers in the reference group in the period of corporate growth and then their hazard is only 10.9% lower than that of reference group workers. Therefore we conclude that evaluation scores become more important for the determination of separation rates of blue-collar workers in the period of downsizing. Such an effect is not found for white-collar workers. The impact of performance evaluation scores on their separation rates does not change significantly across periods and, if anything, relative separation probabilities for differently evaluated workers are more similar in the second period (Table 6.3d).

Workers who have been promoted or have climbed up in the wage scale of the firm have lower separation hazards than those who have not made career from 1987 onwards. Variables that control for transitions in wage scales tend to have a stronger and statistically significant effect in the period of downsizing. In this period, a worker who has climbed up two wage scales, faces a job separation hazard that is permanently 15.7% lower than that of a worker who is still in the same scale as in 1987 (see Table 6.1d, column 3). Moreover, the effect of a recent transition in wage scales is larger and more significant in the period of downsizing than in the period of growth.

Table 6.2d reveals that recent transitions in wage scales especially reduce separation rates of blue-collar workers during downsizing; this effect is less strong for white-collar workers. Moreover, upward transitions in wage scales have a strong and significant negative persistent effect on separation rates for blue-collar workers in the period of workforce expansion, while the persistent effect of a transition is smaller in the period of downsizing. The persistent effect of upward transitions is insignificant for white-collar workers in both periods, yet moving down the wage scale has a larger positive effect and moving up in the wage scale has a negative effect on separation rates in the period of downsizing (Table 6.3d).

Promotions recorded in the wage files reduce the hazard of job separation: The coefficient estimate in column 1 of Table 6.1d is statistically significant at the 1%-level and suggests that the hazard rate falls by 15.8% per promotion. Yet the effect of a promotion differs for different groups of workers and across periods. It is strongest, but only significant at the 5%-level for white-collar workers during the period of corporate growth when a promotion lowers the hazard of job separation by 42.9% (Table 6.3d, column2). But a promotion does not affect the separation hazard of white-collar workers significantly during downsizing. The change of the effect of promotions on separation rates is significant as the p-value in column 4 of Table 6.3d indicates. Interestingly, the opposite holds for blue-collar workers. Promotions do not impact significantly on their separation rates during the growth phase, but reduce the risk of job loss significantly by 15.4% during downsizing. Wage growth associated with promotions is significant only for white-collar workers during downsizing, when 1% extra wage growth leads to a reduction of the separation hazard of 2%.

The completion of a degree affects only the separation rates of blue-collar workers significantly during the period of downsizing, when it raises separation rates by 66.9%. Wage growth associated with a degree completion tends to reduce separation rates, but the effect is insignificant and generally small. It is largest, but insignificant, for blue-collar workers during downsizing when a degree completion has a large positive effect. Then, a percentage point wage increase associated with a degree completion reduces the hazard rate by almost 8% suggesting more a wage increase of more than 8% is required to offset the effect of obtaining a higher degree. Clearly, the wage growth required by blue-collar workers to leave separation rates unaffected after obtaining a higher degree is larger during downsizing, indicating that workers then demand a larger share of the current surplus.

It seems that blue-collar workers make career by increasing their human capital and obtaining degrees and thereby climb the wage scale of the firm, while the career path is via promotions for white-collar workers. The transition to a higher wage scale and wage growth associated with either degree completion or promotion has a stronger negative effect on separation rates for blue-collar workers in the period of downsizing, while the event of obtaining a higher vocational or general schooling degree has a stronger positive effect on separation rates during downsizing. This implies that in periods when prospects are good, most of the reduction in separation probabilities comes from better prospects on future surpluses, while during periods of downsizing, blue-collar workers that make career, e.g. by obtaining a higher degree, require more current wage growth to prevent a separation. When prospects are bad, workers put much weight on their current share of the surplus.

Similarly, promoted white-collar workers are less likely to separate in

the period of expansion, holding constant the wage. A rise into a higher wage scale or wage growth associated with the promotion does not have a significant impact on separation rates in times of expansions when prospects are good. However, the negative effect of a promotion, holding constant the wage, disappears in the period of downsizing, while higher wages have a stronger negative impact on separation rates.

6.6 Effects of Marital Status, Gender, and Worker Status

Married workers face 39.6% lower separation hazards than non-married workers during the period of expansion but only 16.6% lower hazards during the period of downsizing (see Table 6.1e). These effects are quantitatively similar for blue-collar and withe-collar workers (see Tables 6.2e and 6.3e). The differences between periods are always significant (see column 4 of Tables 6.1e, 6.2e and 6.3e).

Men are less likely to separate during the period of expansion (37.3% lower separation hazards than women) but face a higher risk of job loss in the period of downsizing when their job separation hazard is on average about 9.9% higher than that of women (Table 6.1e). Again the difference is highly significant. A comparison of the coefficient estimates of the gender effect in Tables 6.2e and 6.3e reveals, however, that differences of the effect between the episodes of corporate growth and decline are more pronounced and significant only for blue-collar workers. In fact, male white-collar workers are always less likely to separate than female white-collars and there is no significant difference in the point estimate of the gender effect in the two periods; yet the gender effect is only significant among white-collar workers during downsizing (the hazard is then 26.9% lower for white-collar males).

Hazard rates for part-time workers are significantly higher than for fulltime workers. The estimates for the entire sample (Table 6.1e) indicate that hazard rates are 60% higher during expansion and 50% higher during downsizing for part-time workers, but the difference between periods is not significant. Tables 6.2e and 6.3e make however clear that the effect is stronger among blue-collar workers (70% higher hazards for part-timers during expansion and 55% during contraction and no significant difference between periods). While the impact of working part-time is significant for blue-collar
workers in both regime, no significant effect is found for white-collar workers in the period of expansion. The point estimate is even negative then. However, the impact of having part-time status on separation rate changes significantly for white-collar workers during downsizing when part-time employees face 33.6% higher separation hazards.

Workers who are rehired during the observation period face a lower risk of job separation. This may result if some specific investments that were made during the unobserved employment spell are still valuable for the current surplus. While the effect of a previous employment spell with the firm never changes significantly across periods, it is significant only in the second period.

Finally, blue-collar workers have lower separation rates than white-collar workers when the workforce grows, but become more likely to separate during downsizing. This effect which shows up in the different baseline hazards for the sub-samples of blue- and white-collar workers is also reflected by the estimated coefficient of controls for the first observed wage scale of a worker in Table 6.1e. These estimated parameters suggest that separation rates are lowest for workers in the lower wage scales during expansion, but highest during downsizing. While estimates for all but one blue-collar wage scale are negative during expansion and all become positive during downsizing, the change in separation rates is especially pronounced and significant in scales 4-8. Analogously, workers in higher white-collar scales (15-19) are more likely to separate during expansion than workers in the reference scale 11, but all white-collar workers become less likely to separate during the period of expansion. The change is highly significant for workers in scales 15-19 (see column 4 of Table 6.1e).

Moreover, separation rates fall with an increase the first observed wage scale during downsizing. Hence the higher the wage - and therefore the higher the job level - the less likely a worker is to separate in the period of downsizing. This result is supported by the observation that a transition to a higher wage scale reduces the separation rate in the that period. In the period of expansion, separation rates differ less across wage scales. Neither workers in lower wage scales than the reference category (scale 11) nor workers in higher scales have separation rates that differ significantly from those of workers in scale 11. However, workers in the highest scales (15-19) are more likely to separate than those in lower scales (3-9).

7 Conclusion

This paper investigates differences in effects of worker separation characteristics between periods of workforce expansion and contraction. In a downsizing phase, workers are less willing to accept wage incentive contracts that postpone earnings, and career paths within the firm become an increasingly important indicator of a worker's mobility hazard. At the same time, the worker's <u>current</u> share of the surplus becomes more important when the firm's prospects worsen. Workforce reductions thus increase a worker's rate of discounting future earnings at the firm, decreasing the value of postponing mobility. This is in accordance with the implications from theoretical models that take account of the option value associated with a separation. A downsizing firm should recognize this and should offer sufficient current compensation to workers it it wants to stay.

Performance evaluation scores gain importance for blue-collar workers in periods of downsizing. In our data, this effect is not found for white-collar workers. Less than 2% of all workers get performance evaluation score in the lower third of the scale, but they face substantially higher job separation hazards. When turnover rates are high, as in periods of downsizing, individual separation hazards are much lower for those in the upper four deciles of the performance evaluation score distribution.

Career paths within the firm exhibit a dependence relation with a worker's turnover probability. Promotions are important to lower white-collar worker turnover during prosperous periods of the firm, but do not matter in bad times. Blue-collar workers experience the opposite effect of career paths. Promotions do not change the probability of turnover when the workforce expands, but reduces the chance of leaving when the firm is contracting.

Fitting a quadratic age effect produces misleading results. In fact, the effect of age on separation rates is marked by important discontinuities, which are probably driven by discontinuous changes of the value of the retirement option. Option values for retirement schemes change dramatically with the prospects of the firm. Workers kill the option to retire early more frequently in periods of downsizing. However, if early retirement is offered only beyond a certain age, the option to retire early at a given age has important spill-over effects for slightly younger workers who become consequently less likely to separate. A downsizing firm should take this crowding out effect into account when designing early retirement plans. The role of retirement and its relation

to worker turnover rates deserves a more careful treatment in future work on corporate downsizing and worker mobility.

We have presented a case history analysis of worker mobility and compared the characteristics of turnover between prosperous and diminishing periods of a firm. This work can be expanded upon in a variety of ways. First, the behavior of other declining firms could be studied - nothing guarantees that in all such firms turnover will be characterized in the same way as we did. A second, more important route would be the development of a theoretical model to explain differences in turnover behavior between blue- and white-collar workers in good and bad times. The generally important point we have made - that smooth functional forms of the age effect in mobility studies produce misleading results - seems applicable to better understand and design early retirement plans and incentive schemes for corporate reorganizations.

References

- Baker, George, Michael Gibbs, and Bengt Holmstrom (1994). "The Wage Policy of A Firm", *Quarterly Journal of Economics*, vol.109, pp. 921-955.
- Cox, David R. (1972). "Regression Models and Life Tables", *Journal of the Royal Statistical Society*, series B 34, 187-220.

Cox, David R. (1975). "Partial Likelihood", Biometrika, 62, 269-276.

Deterink, A.A.M., B.F.M. Knüppe, A.L. Leuftink and R.J. Schimmelpenninck (1997). *"Bankruptcy Trustees' Investigation of the Causes of the Bankruptcy of N.V. Koninklijke Nederlandse Vliegtuigfabriek Fokker, Fokker Aircraft B.V. and Fokker Admininstration B.V."*

- Dustmann, Christian and Costas Meghir (1998). "Wages, experience and seniority", IFS Working Paper W99/1.
- Hall, Robert E. and Edward P. Lazear (1984). "The excess sensitivity of layoffs and quits to demand", *Journal of Labor Economics*, 2, 223-257.
- Hamermesh, Dan S. and Gerard A. Pfann (1996). "Adjustment Costs in Factor Demand", Journal of Economic Literature, 34, 1264-1292.
- Imbens, Guido W. (1994). "Transition Models in a Non-Stationary Environment", *Review of Economics and Statistics*, 76(4), 703-720.
- Jovanovic, Boyan (1979a). "Firm-specific Capital and Turnover", *Journal of Political Economy*, 87, 1246-1260.
- Jovanovic, Boyan (1979b). "Job matching and the theory of turnover", *Journal of Political Economy*, 87, 972-990.
- Kuhn, P. (1988). "A nonuniform pricing model of union wages and employment", Journal of Political Economy, 96, 473-508.

Kuhn, P. and J. Roberts (1989). "Seniority and distribution in a two-worker trade union",

Quarterly Journal of Economics, 104, 485-505.

Lazear, Edward P. (1979). "Why is there mandatory retirement", *Journal of Political Economy*, 87, 1261-1264.

Lazear, Edward P. (1995). Personnel Economics, MIT Press.

- MacLeod, W.B. and J.M. Malcomson (1993). "Investment, holdups, and the form of market contracts", *American Economic Review*, 83, 811-837.
- Pfann, Gerard (1998). *Options to Quit*, METEOR Working Paper, no. RM/98/016, Maastricht.
- Stock, James H. and David A. Wise (1990). "Pensions, the Option Value of Work, and Retirement", *Econometrica*, 58(5). 131-159.
- Teulings, Coen and Joop Hartog (1998). *Corporatism or Competition?*, Cambridge University Press, Cambridge.

Table 6.1a

Tenure

1	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
ten3m	3.452	3.367	3.100	0.0374
	(0.083)**	(0.153)**	(0.128)**	
ten6m	1.800	1.459	1.949	0.0001
	(0.090)**	(0.163)**	(0.128)**	
ten9m	1.523	1.175	1.744	0.0000
	(0.088)**	(0.161)**	(0.121)**	
ten12m	1.815	1.561	1.897	0.0010
	(0.081)**	(0.160)**	(0.102)**	
ten1	1.495	0.926	1.931	0.0000
	(0.067)**	(0.145)**	(0.078)**	
ten2	1.493	1.028	1.803	0.0000
	(0.065)**	(0.145)**	(0.075)**	
ten3	1.408	1.074	1.627	0.0000
	(0.064)**	(0.147)**	(0.072)**	
ten4	1.100	0.918	1.252	0.0000
	(0.065)**	(0.155)**	(0.073)**	
ten5_9	0.780	0.745	0.823	0.2149
	(0.056)**	(0.133)**	(0.063)**	
ten10_14	0.341	0.158	0.419	0.0001
	(0.058)**	(0.143)	(0.065)**	
ten20_24	-0.108	-0.066	-0.125	0.5242
	(0.082)	(0.186)	(0.093)	
ten25_29	-0.094	0.174	-0.174	0.0010
	(0.088)	(0.174)	(0.105)*	
ten30_34	0.291	0.242	0.281	0.6752
	(0.077)**	(0.154)	(0.093)**	
ten35_39	0.359	0.079	0.442	0.0001
	(0.079)**	(0.158)	(0.094)**	
ten40	0.976	0.642	1.132	0.0000
	(0.093)**	(0.175)**	(0.114)**	
Observations	172492	97806	87492	

Table 6.1b

Age Effects

	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
old17	-0.231	0.038	0.099	0.9324
	(0.418)	(0.528)	(0.719)	
old18	-0.337	-0.460	0.266	0.0009
	(0.167)*	(0.276)*	(0.218)	
old19	-0.321	-0.478	0.026	0.0010
	(0.127)**	(0.236)*	(0.153)	
old20	-0.043	0.002	0.188	0.1455
	(0.105)	(0.197)	(0.128)	
old21	0.070	0.275	0.147	0.2817
	(0.096)	(0.182)	(0.119)	
old22	0.041	0.508	-0.132	0.0000
	(0.092)	(0.172)**	(0.118)	
old23	0.188	0.621	0.005	0.0000
	(0.089)*	(0.169)**	(0.111)	
old24	0.067	0.438	-0.087	0.0000
	(0.086)	(0.169)**	(0.104)	
old25	0.295	0.534	0.214	0.0008
	(0.081)**	(0.167)**	(0.095)*	
old26	0.162	0.494	0.051	0.0000
	(0.082)*	(0.167)**	(0.096)	
old27	0.188	0.373	0.135	0.0108
	(0.081)*	(0.170)*	(0.093)	
old28	0.117	0.414	0.010	0.0000
	(0.082)	(0.169)**	(0.095)	
old29	0.129	0.412	0.025	0.0000
	(0.082)	(0.170)**	(0.095)	
old30	0.152	0.432	0.041	0.0000
	(0.082)*	(0.170)**	(0.094)	
old31	0.086	0.344	-0.003	0.0003
-	(0.084)	(0.175)*	(0.097)	
old32	0.156	0.495	0.054	0.0000
	(0.084)*	(0.175)**	(0.097)	
old33	-0.020	0.184	-0.076	0.0125
	(0.091)	(0.189)	(0.104)	
old34	-0.056	0.102	-0.097	0.0565
	(0.092)	(0.196)	(0.104)	
old36	-0.014	0.145	-0.059	0.0587
	(0.095)	(0.199)	(0.108)	
old37	0.014	-0.081	0.049	0.2386
	(0.098)	(0.213)	(0.111)	
old38	-0.107	-0.232	-0.085	0.2099
	(0.104)	(0.229)	(0.117)	
old39	-0.116	-0.378	-0.035	0.0031
	(0.104)	(0.241)	(0.116)	
old40	0.081	0.229	0.046	0.1205
	(0.102)	(0.209)	(0.118)	

(continued on next page)

Table 6.1b (continued)

li -	ii.		[Ϋ́.	
old41	-0.140	-0.336	-0.037	0.0200	
	(0.114)	(0.251)	(0.128)		
old42	-0.280	0.139	-0.425	0.0001	
	(0.120)**	(0.222)	(0.145)**		
old43	-0.182	-0.456	-0.081	0.0037	
	(0.116)	(0.273)*	(0.129)		
old44	-0.308	-0.135	-0.326	0.1987	
	(0.129)**	(0.264)	(0.149)*		
old45	-0.252	-0.231	-0.214	0.9112	
	(0.132)*	(0.291)	(0.148)		
old46	-0.314	-0.390	-0.284	0.4808	
	(0.135)**	(0.315)	(0.150)*		
old47	-0.499	-0.484	-0.465	0.9081	
	(0.146)**	(0.336)	(0.163)**		
old48	-0.535	-0.433	-0.506	0.6776	
	(0.155)**	(0.336)	(0.175)**		
old49	-0.787	-0.258	-0.925	0.0014	
	(0.173)**	(0.315)	(0.209)**		
old50	-1.109	-1.181	-1.062	0.6510	
	(0.229)**	(0.472)**	(0.262)**		
old51	-1.744	-1.127	-2.000	0.0238	
	(0.297)**	(0.473)**	(0.386)**		
old52	-1.712	-1.262	-1.871	0.0936	
	(0.297)**	(0.523)**	(0.363)**		
old53	-1.350	-1.966	-1.201	0.0082	
	(0.268)**	(0.724)**	(0.289)**		
old54	-1.151	-0.390	-1.402	0.0001	
	(0.205)**	(0.352)	(0.256)**		
old55	1.878	-0.912	2.255	0.0000	
	(0.092)**	(0.437)*	(0.101)**		
old56	1.326	-0.230	1.873	0.0000	
	(0.112)**	(0.319)	(0.122)**		
old57	1.521	0.264	2.019	0.0000	
	(0.110)**	(0.279)	(0.122)**		
old58	1.476	-0.035	2.223	0.0000	
	(0.114)**	(0.304)	(0.125)**		
old59	0.683	0.518	0.867	0.0444	
	(0.143)**	(0.257)*	(0.174)**	010111	
old60	3 543	3 836	3 317	0.0000	
	(0.086)**	(0.169)**	(0.107)**	0.0000	
old61	1 618	2.183	1 259	0.0000	
01001	(0.154)**	(0.243)**	(0.216)**	0.0000	
old62	1 840	2 358	1 554	0.0002	
01002	(0.161)**	(0.261)**	(0.214)**	0.0002	
01d63	0.366	1 971	_0 39/	0.000	
	(0.252)	(0 359)**	(0.370)	0.0000	
old64	1 457	2 750	1 122	0.0000	
	(0.136)**	(0.240)**	(0.168)**	0.0000	
	(0.150)	(0.2+0)	(0.100)		
Observations	172492	97806	87492		

Table 6.1c

Education

	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
school_mis	0.364	0.645	0.253	0.0000
	(0.037)**	(0.071)**	(0.044)**	
school_1	0.298	0.325	0.310	0.9060
	(0.096)**	(0.163)*	(0.124)**	
school_2	0.249	0.222	0.272	0.2577
	(0.038)**	(0.076)**	(0.044)**	
school_3	0.192	0.325	0.173	0.0106
	(0.049)**	(0.090)**	(0.059)**	
school_5	0.297	0.596	0.178	0.0000
	(0.070)**	(0.117)**	(0.089)*	
school_6	0.103	0.238	0.055	0.0002
	(0.042)**	(0.080)**	(0.049)	
school_7	0.208	0.528	0.056	0.0000
	(0.075)**	(0.124)**	(0.096)	
school_8	0.112	0.415	0.002	0.0000
	(0.057)*	(0.101)**	(0.071)	
school_9	0.210	0.459	0.106	0.0000
	(0.067)**	(0.117)**	(0.083)	
Observations	172492	97806	87492	

Table 6.1d

	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
eval_mis	0.348	0.619		-
	(0.137)**	(0.140)**		
eval_1	1.837	1.813	1.946	0.5049
	(0.142)**	(0.220)**	(0.199)**	
eval_2	0.677	0.847	0.573	0.0001
	(0.055)**	(0.088)**	(0.070)**	
eval_4	-0.225	-0.122	-0.279	0.0000
	(0.027)**	(0.052)**	(0.031)**	
eval_5	-0.361	-0.112	-0.449	0.0000
	(0.049)**	(0.085)	(0.060)**	
eval_6	-0.350	-0.003	-0.472	0.0002
	(0.096)**	(0.150)	(0.127)**	
∆scale1	-0.759	-0.354	-0.844	0.0000
	(0.046)**	(0.099)**	(0.052)**	
∆scale2	-0.686	-0.011	-0.781	0.0000
	(0.136)**	(0.264)	(0.163)**	
∆scale<0	-0.646	0.186	-0.768	0.0000
	(0.195)**	(0.543)	(0.210)**	
d_scale<0	0.105	-0.010	0.195	0.1319
	(0.123)	(0.320)	(0.136)	
d_scale1	-0.042	-0.192	-0.040	0.0000
	(0.031)	(0.070)**	(0.037)	
d_scale2	-0.156	-0.272	-0.171	0.0285
	(0.041)**	(0.125)*	(0.046)**	
d_scale3	-0.127	-0.222	-0.160	0.2841
	(0.052)**	(0.197)	(0.058)**	
Promotion	-0.147	-0.206	-0.162	0.3932
	(0.049)**	(0.142)	(0.052)**	
prom_∆wage	0.018	0.017	0.030	0.1206
	(0.008)**	(0.023)	(0.008)**	
Degree	0.479	-0.085	0.544	0.0000
	(0.101)**	(0.238)	(0.121)**	
degr_∆wage	-0.007	0.102	-0.056	0.0000
	(0.030)	(0.058)*	(0.038)	
Observations	172492	97806	87492	

Performance Evaluation and Career Paths

Table 6.1e

Marital Status, gender and work status

	1			
	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
married	-0.298	-0.505	-0.182	0.0000
	(0.023)**	(0.040)**	(0.029)**	
divorced	-0.308	-0.830	-0.082	0.0000
	(0.106)**	(0.222)**	(0.121)	
male	-0.126	-0.467	0.094	0.0000
	(0.034)**	(0.056)**	(0.044)*	
part_time	0.390	0.474	0.407	0.2075
	(0.044)**	(0.081)**	(0.053)**	
rehire	-0.358	-0.128	-0.433	0.0018
	(0.086)**	(0.185)	(0.098)**	
scale_2	0.129	0.699	0.099	0.1115
	(0.259)	(0.359)*	(0.377)	
scale_3	-0.137	-0.297	0.018	0.0903
	(0.139)	(0.222)	(0.186)	
scale_4	0.140	-0.207	0.424	0.0000
	(0.085)	(0.144)	(0.108)**	
scale_5	0.230	-0.067	0.490	0.0000
	(0.081)**	(0.137)	(0.104)**	
scale_6	0.195	-0.097	0.418	0.0000
	(0.080)**	(0.134)	(0.102)**	
scale_7	0.129	-0.042	0.265	0.0030
	(0.081)	(0.132)	(0.104)**	
scale_8	0.122	-0.129	0.277	0.0001
	(0.080)	(0.132)	(0.103)**	
scale_9	0.086	-0.080	0.239	0.0036
	(0.086)	(0.142)	(0.110)*	
scale_10	0.069	0.065	0.105	0.7334
	(0.093)	(0.153)	(0.118)	
scale_12	-0.432	-0.208	-0.476	0.0153
	(0.086)**	(0.142)	(0.110)**	
scale_13	-0.205	-0.113	-0.193	0.5087
	(0.096)*	(0.161)	(0.121)	
scale_14	-0.306	-0.101	-0.328	0.0685
	(0.098)**	(0.161)	(0.125)**	
scale_15	-0.175	0.197	-0.402	0.0000
	(0.102)*	(0.159)	(0.136)**	
scale_16	-0.208	0.216	-0.412	0.0000
	(0.111)*	(0.174)	(0.146)**	
scale_17	-0.130	0.287	-0.380	0.0002
	(0.141)	(0.226)	(0.181)*	
scale_18	-0.373	0.332	-0.705	0.0000
	(0.139)**	(0.213)	(0.187)**	
scale_19	-0.628	0.315	-1.068	0.0000
	(0.232)**	(0.344)	(0.318)**	
Observations	172492	97806	87492	

Table 6.1f:

Location

	(1)	(2)	(3)	
	1987-1996	1987-1991	1991-1996	p-value
location1	1.179	0.689		-
	(0.271)**	(0.278)**		
location2	0.202	0.287	0.126	0.0000
	(0.031)**	(0.056)**	(0.038)**	
location3	-0.537	-0.052	-0.818	0.0000
	(0.076)**	(0.112)	(0.106)**	
location4	-0.216	-0.117	-0.253	0.0329
	(0.052)**	(0.091)	(0.064)**	
location5	-0.019	-0.142	-0.014	0.0422
	(0.054)	(0.105)	(0.063)	
location6	0.136	0.270	0.047	0.0000
	(0.036)**	(0.073)**	(0.043)	
location7	-0.532	0.008	-0.857	0.0000
	(0.053)**	(0.090)	(0.067)**	
Observations	172492	97806	87492	

Table 6.2a

Tenure

Blue-Collar Workers						
	1987-1996	1987-1991	1991-1996	p-value		
ten3m	3.506	3.360	3.053	0.0356		
	(0.092)**	(0.178)**	(0.146)**			
ten6m	1.819	1.381	1.907	0.0002		
	(0.100)**	(0.187)**	(0.143)**			
ten9m	1.581	1.104	1.812	0.0000		
	(0.096)**	(0.185)**	(0.131)**			
ten12m	1.850	1.490	1.884	0.0004		
	(0.090)**	(0.185)**	(0.112)**			
ten1	1.555	0.764	2.036	0.0000		
	(0.075)**	(0.170)**	(0.087)**			
ten2	1.517	0.838	1.846	0.0000		
	(0.073)**	(0.170)**	(0.082)**			
ten3	1.414	0.879	1.638	0.0000		
	(0.071)**	(0.172)**	(0.079)**			
ten4	1.075	0.750	1.206	0.0000		
	(0.072)**	(0.183)**	(0.080)**			
ten5_9	0.673	0.555	0.674	0.0870		
	(0.062)**	(0.154)**	(0.070)**			
ten10_14	0.343	0.073	0.405	0.0000		
	(0.064)**	(0.164)	(0.071)**			
ten20_24	-0.188	-0.230	-0.220	0.9255		
	(0.097)*	(0.229)	(0.108)*			
ten25_29	-0.041	0.094	-0.132	0.0719		
	(0.103)	(0.204)	(0.125)			
ten30_34	0.375	0.163	0.376	0.0601		
	(0.091)**	(0.181)	(0.113)**			
ten35_39	0.300	-0.021	0.351	0.0013		
<u></u>	(0.095)**	(0.187)	(0.116)**			
ten40	0.952	0.667	1.100	0.0027		
 	(0.116)**	(0.217)**	(0.144)**			
Observations	125465	73771	61118			

Table 6.2b

Age

$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old17 -0.166 -0.089 0.245 0.6437 (0.421) (0.538) (0.722) - old18 -0.295 -0.594 0.351 0.0000 $(0.174)^*$ $(0.225)^*$ (0.225) - old19 -0.292 -0.598 0.034 0.0001 $(0.135)^*$ $(0.258)^*$ (0.162) - old20 -0.017 -0.104 0.162 0.0542 (0.114) (0.222) (0.138) - - old21 0.089 0.171 0.118 0.6809 (0.106) (0.209) (0.129) - - old22 0.059 0.409 -0.144 0.0000 (0.103) $(0.201)^*$ (0.127) - - old23 0.174 0.511 -0.016 0.0001 (0.098) $(0.200)^*$ (0.116) - old24 0.055 0.366 -0.103 0.0021 $(0.093)^{**}$ <
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old18 -0.295 -0.594 0.351 0.0000 $(0.174)^*$ $(0.295)^*$ (0.225) 0001 $0ld19$ -0.292 -0.598 0.034 0.0001 $(0.135)^*$ $(0.258)^*$ (0.162) 0001 $0ld20$ -0.017 -0.104 0.162 0.0542 (0.114) (0.222) (0.138) 0000 (0.106) (0.209) (0.129) 00122 $0ld21$ 0.089 0.171 0.118 0.6809 (0.106) (0.209) (0.129) 00000 (0.103) $(0.201)^*$ (0.127) 00122 $0ld23$ 0.174 0.511 -0.016 0.0000 $(0.103)^*$ $(0.199)^{**}$ $(0.122)^*$ 00124 0.055 0.366 -0.103 0.0001 $(0.098)^*$ $(0.200)^*$ $(0.116)^*$ 0.0271 0.0271 $0ld24$ 0.055 0.366 0.117 0.0271 0.0271
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old19 -0.292 -0.598 0.034 0.0001 $(0.135)^*$ $(0.258)^*$ (0.162) old20 -0.017 -0.104 0.162 (0.114) (0.222) (0.138) old21 0.089 0.171 0.118 0.6809 (0.106) (0.209) (0.129) old22 0.059 0.409 -0.144 0.0000 (0.103) $(0.201)^*$ (0.127) old23 0.174 0.511 -0.016 0.0000 $(0.100)^*$ $(0.199)^{**}$ (0.122) old24 0.055 0.366 -0.103 0.0001 (0.098) $(0.200)^*$ (0.116) old25 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ old26 0.164 0.361 0.117 0.0271 $(0.096)^*$ $(0.201)^*$ $(0.108)^*$ 0.0015 $(0.095)^{**}$ (0.204) $(0.108)^*$ 0.0015 $(0.096)^*$ $(0.203)^*$ (0.111) 0.0971 (0.097) (0.211) (0.110) 0.0003 (0.097) $(0.208)^*$ (0.111) 0.0003 (0.097) (0.220) (0.113) 0.0003 (0.097) (0.220) (0.113) 0.0003 (0.097) (0.220) (0.113) $(0.103)^*$ (0.100) (0.220) (0.113) $(0.003)^*$ (0.097) (0.220) (0.113) $(0.202)^*$ (0.097) $(0.203)^*$ <t< td=""></t<>
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old20 -0.017 -0.104 0.162 0.0542 (0.114) (0.222) (0.138) 0 old21 0.089 0.171 0.118 0.6809 (0.106) (0.209) (0.129) 0 old22 0.059 0.409 -0.144 0.0000 (0.103) (0.201)* (0.127) 0 old23 0.174 0.511 -0.016 0.0000 (0.100)* (0.199)** (0.122) 0 0 old24 0.055 0.366 -0.103 0.0001 (0.098) (0.200)* (0.116) 0 0 old25 0.338 0.509 0.283 0.0358 (0.093)** (0.198)** (0.108)** 0 0 old26 0.164 0.361 0.117 0.0271 (0.096)* (0.201)* (0.108)* 0 0 old27 0.222 0.325 0.209 0.2819
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old21 0.089 0.171 0.118 0.6809 (0.106) (0.209) (0.129) old22 0.059 0.409 -0.144 0.0000 (0.103) $(0.201)^*$ (0.127) old23 0.174 0.511 -0.016 0.0000 $(0.100)^*$ $(0.199)^{**}$ (0.122) old24 0.055 0.366 -0.103 0.0001 (0.098) $(0.200)^*$ (0.116) old25 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ old26 0.164 0.361 0.117 0.0271 $(0.096)^*$ $(0.201)^*$ $(0.108)^*$ old27 0.222 0.325 0.209 0.2819 $(0.095)^{**}$ (0.204) $(0.108)^*$ old28 0.112 0.373 0.022 0.0015 (0.096) $(0.203)^*$ (0.111) old29 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.111) 0.0003 (0.097) $(0.208)^*$ (0.111) old31 0.063 0.129 0.027 0.3645
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old22 0.059 0.409 -0.144 0.0000 (0.103) $(0.201)^*$ (0.127) old23 0.174 0.511 -0.016 0.0000 $(0.100)^*$ $(0.199)^{**}$ (0.122) old24 0.055 0.366 -0.103 0.0001 (0.098) $(0.200)^*$ (0.116) old25 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ old26 0.164 0.361 0.117 0.0271 $(0.096)^*$ $(0.201)^*$ (0.110) old27 0.222 0.325 0.209 0.2819 $(0.095)^{**}$ (0.204) $(0.108)^*$ old28 0.112 0.373 0.022 0.0015 (0.096) $(0.203)^*$ (0.111) old29 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.111) old30 0.128 0.411 0.015 0.0003 (0.097) $(0.208)^*$ (0.111) old31 0.063 0.129 0.027 0.3645
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old23 0.174 0.511 -0.016 0.0000 $(0.100)^*$ $(0.199)^{**}$ (0.122) old24 0.055 0.366 -0.103 0.0001 (0.098) $(0.200)^*$ (0.116) old25 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ old26 0.164 0.361 0.117 0.0271 $(0.096)^*$ $(0.201)^*$ (0.110) old27 0.222 0.325 0.209 0.2819 $(0.095)^{**}$ (0.204) $(0.108)^*$ old28 0.112 0.373 0.022 0.0015 (0.096) $(0.203)^*$ (0.111) old29 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.110) old30 0.128 0.411 0.015 0.0003 (0.097) $(0.208)^*$ (0.111) old31 0.063 0.129 0.027 0.3645
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
$old24$ 0.055 0.366 -0.103 0.0001 (0.098) $(0.200)^*$ (0.116) $old25$ 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ $old26$ 0.164 0.361 0.117 0.0271 $(0.096)^*$ $(0.201)^*$ (0.110) $old27$ 0.222 0.325 0.209 0.2819 $(0.095)^{**}$ (0.204) $(0.108)^*$ $old28$ 0.112 0.373 0.022 0.0015 (0.096) $(0.203)^*$ (0.111) $old29$ 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.110) $old30$ 0.128 0.411 0.015 0.0003 (0.097) $(0.208)^*$ (0.111) $old31$ 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0002
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
$old25$ 0.338 0.509 0.283 0.0358 $(0.093)^{**}$ $(0.198)^{**}$ $(0.108)^{**}$ $0.108)^{**}$ $old26$ 0.164 0.361 0.117 0.0271 $(0.096)^{*}$ $(0.201)^{*}$ (0.110) 0.0271 $old27$ 0.222 0.325 0.209 0.2819 $(0.095)^{**}$ (0.204) $(0.108)^{*}$ 0.021 $old28$ 0.112 0.373 0.022 0.0015 (0.096) $(0.203)^{*}$ (0.111) 0.0971 $old29$ 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.110) 0.0003 (0.097) $(0.208)^{*}$ (0.111) $old31$ 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0003
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
old28 0.112 0.373 0.022 0.0015 (0.096) (0.203)* (0.111) 0.0971 old29 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.110) 0.0003 old30 0.128 0.411 0.015 0.0003 (0.097) (0.208)* (0.111) 0.015 0.0003 old31 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0003
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
old29 0.081 0.220 0.037 0.0971 (0.097) (0.211) (0.110) 0.0003 old30 0.128 0.411 0.015 0.0003 (0.097) (0.208)* (0.111) 0.015 0.0003 old31 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0003
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
old30 0.128 0.411 0.015 0.0003 (0.097) (0.208)* (0.111) 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0000 0.0000 0.0000
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
old31 0.063 0.129 0.027 0.3645 (0.100) (0.220) (0.113) 0.0000
(0.100) (0.220) (0.113)
01032 0.028 0.270 -0.034 0.0089
(0.103) (0.221) (0.116)
old33 -0.0/4 0.152 -0.154 0.0256
(0.110) (0.232) (0.125)
01034 -0.113 -0.180 -0.100 0.5153
(0.110) (0.257) (0.122)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
(0.110) (0.233) (0.130)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
(0.127) (0.301) (0.141)
(0.124) $(0.307)*$ (0.137) $(0.0002$
0.124 0.224 0.271 0.221 0.7072
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Standard errors in parentheses * significant at 5% level; ** significant at 1% level

(continued on next page)

Table 6.2b (continued)

old41	-0.091	-0.482	0.045	0.0005
	(0.137)	(0.321)	(0.151)	
old42	-0.258	0.052	-0.362	0.0660
	(0.142)*	(0.274)	(0.169)*	
old43	-0.182	-0.635	-0.063	0.0002
	(0.139)	(0.351)*	(0.152)	
old44	-0.298	-0.149	-0.308	0.3550
	(0.150)*	(0.314)	(0.172)*	
old45	-0.210	-0.227	-0.163	0.7102
	(0.155)	(0.351)	(0.173)	
old46	-0.335	-0.655	-0.272	0.0257
	(0.158)*	(0.419)	(0.172)	
old47	-0.401	-0.376	-0.406	0.8717
	(0.167)**	(0.379)	(0.186)*	
old48	-0.567	-0.737	-0.500	0.2620
	(0.190)**	(0.446)*	(0.211)**	
old49	-0.923	-0.708	-0.943	0.3332
	(0.213)**	(0.447)	(0.243)**	
old50	-0.958	-1.140	-0.901	0.3982
	(0.250)**	(0.532)*	(0.283)**	
old51	-1.945	-1.289	-2.243	0.0608
	(0.387)**	(0.606)*	(0.509)**	
old52	-2.045	-1.216	-2.502	0.0278
	(0.417)**	(0.607)*	(0.585)**	
old53	-1.245	-2.295	-1.050	0.0001
	(0.301)**	(1.017)*	(0.316)**	
old54	-1.277	-0.257	-1.611	0.0000
	(0.244)**	(0.400)	(0.316)**	
old55	1.922	-0.569	2.267	0.0000
	(0.107)**	(0.449)	(0.116)**	
old56	1.486	-0.262	1.983	0.0000
	(0.130)**	(0.383)	(0.141)**	
old57	1.627	0.338	2.098	0.0000
	(0.133)**	(0.321)	(0.147)**	
old58	1.550	-0.141	2.375	0.0000
	(0.139)**	(0.371)	(0.150)**	
old59	0.681	0.501	0.752	0.2750
	(0.179)**	(0.306)	(0.230)**	
old60	3.735	3.888	3.465	0.0009
	(0.102)**	(0.205)**	(0.128)**	
old61	1.592	1.852	1.452	0.1414
	(0.204)**	(0.336)**	(0.272)**	
old62	1.804	2.365	1.541	0.0034
	(0.212)**	(0.342)**	(0.281)**	
old63	-0.782	0.656	-1.280	0.0011
	(0.461)*	(0.737)	(0.593)*	
old64	1.269	2.436	0.986	0.0000
	(0.167)**	(0.301)**	(0.204)**	
Observations	125465	73771	61118	

Table 6.2c

Education

	Blue-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
school_mis	0.343	0.577	0.242	0.0000	
	(0.039)**	(0.075)**	(0.047)**		
school_1	0.370	0.358	0.366	0.9481	
	(0.097)**	(0.163)*	(0.126)**		
school_2	0.257	0.200	0.284	0.0657	
	(0.039)**	(0.078)**	(0.046)**		
school_3	0.199	0.300	0.200	0.1032	
	(0.051)**	(0.093)**	(0.062)**		
school_5	0.327	0.587	0.214	0.0002	
	(0.076)**	(0.123)**	(0.099)*		
school_6	0.145	0.264	0.104	0.0020	
	(0.044)**	(0.084)**	(0.052)*		
school_7	0.371	0.632	0.168	0.0015	
	(0.104)**	(0.156)**	(0.143)		
school_8	0.425	0.497	0.436	0.6509	
	(0.105)**	(0.170)**	(0.135)**		
school_9	0.631	0.004	0.970	0.0010	
	(0.254)**	(0.510)	(0.294)**		
Observations	125465	73771	61118		

Table 6.2d

Performance	evaluation	and	career	paths
-------------	------------	-----	--------	-------

		Blue-Colla	ar Workers	
	1987-1996	1987-1991	1991-1996	p-value
eval_mis	0.448	0.720		-
	(0.148)**	(0.152)**		
eval_1	1.163	1.006	1.260	0.4257
	(0.222)**	(0.329)**	(0.319)**	
eval_2	0.681	0.972	0.517	0.0000
	(0.069)**	(0.116)**	(0.088)**	
eval_4	-0.292	-0.115	-0.362	0.0000
	(0.029)**	(0.059)*	(0.034)**	
eval_5	-0.475	-0.121	-0.569	0.0000
	(0.051)**	(0.092)	(0.063)**	
eval_6	-0.523	-0.025	-0.642	0.0000
	(0.099)**	(0.155)	(0.132)**	
Δ scale1	-0.859	-0.348	-0.988	0.0000
	(0.050)**	(0.109)**	(0.057)**	
∆scale2	-1.344	-0.209	-1.798	0.1128
	(0.709)*	(1.010)	(1.002)*	
Δ scale<0	-0.775	0.190	-0.930	0.0000
	(0.219)**	(0.550)	(0.242)**	
d_scale<0	-0.151	0.112	-0.113	0.1444
	(0.138)	(0.340)	(0.154)	
d_scale1	-0.070	-0.209	-0.049	0.0001
	(0.035)*	(0.078)**	(0.041)	
d_scale2	-0.184	-0.461	-0.124	0.0000
	(0.047)**	(0.151)**	(0.053)**	
d_scale3	-0.090	-0.652	-0.024	0.0000
	(0.062)	(0.349)*	(0.068)	
Promotion	-0.123	-0.068	-0.143	0.3455
	(0.074)*	(0.208)	(0.080)*	
prom_∆wage	-0.007	0.013	-0.001	0.3478
	(0.014)	(0.040)	(0.015)	
Degree	0.427	-0.016	0.512	0.0003
	(0.133)**	(0.450)	(0.145)**	
degr_∆wage	-0.027	0.079	-0.087	0.0008
	(0.046)	(0.168)	(0.050)*	
Observations	125465	73771	61118	

Table 6.2e

	Blue-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
married	-0.289	-0.521	-0.159	0.0000	
	(0.027)**	(0.047)**	(0.033)**		
divorced	-0.314	-0.939	-0.042	0.0000	
	(0.115)**	(0.254)**	(0.130)		
male	-0.048	-0.492	0.217	0.0000	
	(0.038)	(0.062)**	(0.050)**		
part_time	0.444	0.534	0.438	0.1131	
	(0.049)**	(0.088)**	(0.061)**		
rehire	-0.354	-0.076	-0.418	0.0021	
	(0.098)**	(0.208)	(0.111)**		
scale_2	-0.255	0.727	-0.387	0.0040	
	(0.269)	(0.367)*	(0.387)		
scale_3	-0.293	-0.106	-0.387	0.1648	
	(0.149)*	(0.230)	(0.202)*		
scale_4	-0.036	-0.123	0.031	0.2436	
	(0.100)	(0.159)	(0.132)		
scale_5	0.050	0.032	0.059	0.8308	
	(0.096)	(0.150)	(0.127)		
scale_6	-0.012	-0.030	-0.035	0.9706	
	(0.094)	(0.147)	(0.125)		
scale_7	-0.084	0.034	-0.199	0.0613	
	(0.093)	(0.143)	(0.124)		
scale_8	-0.082	-0.057	-0.167	0.3642	
	(0.091)	(0.141)	(0.121)		
scale_9	-0.097	-0.040	-0.151	0.3809	
	(0.096)	(0.150)	(0.127)		
scale_10	-0.117	0.102	-0.289	0.0048	
<u></u>	(0.104)	(0.161)	(0.139)*		
Observations	125465	73771	61118		

Marital Status , gender and work status

Table 6.2f

Location

	Blue-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
location1	1.431	1.276		-	
	(0.503)**	(0.509)**			
location2	0.132	0.238	0.035	0.0000	
	(0.033)**	(0.061)**	(0.040)		
location3	-0.415	0.186	-0.742	0.0026	
	(0.131)**	(0.186)	(0.185)**		
location4	-0.290	-0.182	-0.291	0.1098	
	(0.056)**	(0.098)*	(0.068)**		
location5	-0.147	-0.323	-0.141	0.0102	
	(0.061)**	(0.124)**	(0.071)*		
location6	0.000	0.172	-0.082	0.0000	
	(0.039)	(0.081)*	(0.046)*		
location7	-0.734	-0.024	-1.127	0.0000	
	(0.060)**	(0.098)	(0.077)**		
Observations	125465	73771	61118		

Table 6.3a

Tenure

	White-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
ten3m	3.199	3.107	3.110	0.9926	
	(0.213)**	(0.353)**	(0.297)**		
ten6m	1.676	1.557	1.853	0.3358	
	(0.228)**	(0.373)**	(0.308)**		
ten9m	1.346	1.357	1.291	0.8479	
	(0.232)**	(0.363)**	(0.342)**		
ten12m	1.702	1.735	1.725	0.9712	
	(0.204)**	(0.349)**	(0.268)**		
ten1	1.501	1.488	1.545	0.7805	
	(0.163)**	(0.300)**	(0.204)**		
ten2	1.599	1.628	1.642	0.9416	
	(0.158)**	(0.297)**	(0.193)**		
ten3	1.433	1.636	1.400	0.2132	
	(0.156)**	(0.297)**	(0.189)**		
ten4	1.251	1.392	1.272	0.5295	
	(0.159)**	(0.306)**	(0.190)**		
ten5_9	1.259	1.490	1.223	0.1040	
	(0.139)**	(0.275)**	(0.164)**		
ten10_14	0.586	0.602	0.566	0.8310	
	(0.148)**	(0.306)*	(0.170)**		
ten20_24	0.294	0.328	0.237	0.6541	
	(0.172)*	(0.338)	(0.202)		
ten25_29	0.108	0.558	-0.076	0.0037	
	(0.183)	(0.346)	(0.219)		
ten30_34	0.418	0.355	0.293	0.7558	
	(0.163)**	(0.307)	(0.198)		
ten35_39	0.704	0.348	0.686	0.0936	
	(0.166)**	(0.312)	(0.201)**		
ten40	1.214	0.815	1.417	0.0068	
	(0.180)**	(0.328)**	(0.223)**		
Observations	47027	24035	26374		

Table 6.3b

Age

1 ige		White Coller Workers				
	1097 1006	1097 1001	1001 1006	n volvo		
.1.117	1987-1990	1987-1991	1991-1990	p-value		
old1/				_		
.1.110				_		
old18				_		
1110				_		
01019				_		
.1.120	1.002		1.010	_		
01020	1.892		1.919	-		
-1401	(0.754)**	2.051	(0.785)***	_		
01021	1.830	2.031		-		
-1400	(1.040)*	(1.091)*		_		
01022	0.420	1.122		-		
-1402	(0.728)	(0.773)	0.477	0.0190		
01023	1.120	1.022	0.477	0.0189		
-1424	$(0.277)^{**}$	$(0.400)^{**}$	(0.488)	0.4720		
01024	0.641	0.812	0.600	0.4739		
1.125	(0.226)**	$(0.379)^*$	(0.296)*	0.0210		
01025	0.398	0.765	0.133	0.0210		
-1426	(0.202)*	$(0.340)^{*}$	(0.274)	0.0006		
01026	0.692	1.134	0.395	0.0006		
1.107	(0.172)**	$(0.312)^{**}$	(0.216)*	0.0224		
old27	0.482	0.749	0.315	0.0334		
.1.129	(0.168)**	$(0.316)^{**}$	(0.204)	0.0515		
01028	0.502	0.739	0.364	0.0515		
1.120	(0.161)**	(0.311)**	(0.193)*	0.0002		
old29	0.553	1.013	0.282	0.0002		
1.120	(0.158)**	(0.297)**	(0.193)	0.0461		
01030	0.419	0.6//	0.306	0.0461		
1.12.1	(0.157)**	(0.303)*	(0.186)*	0.0000		
01031	0.269	0.832	-0.014	0.0000		
1.122	(0.158)*	(0.297)**	(0.192)	0.0000		
old32	0.489	0.914	0.313	0.0008		
1.122	(0.151)**	(0.294)**	(0.179)*	0.4442		
old33	0.10/	0.224	0.079	0.4443		
1.10.4	(0.164)	(0.328)	(0.190)	0.0001		
old34	0.079	0.505	-0.106	0.0021		
1126	(0.166)	(0.314)	(0.199)	0.0425		
old36	-0.142	0.169	-0.258	0.0435		
1.107	(0.179)	(0.344)	(0.212)	0.1506		
old37	-0.153	-0.373	-0.076	0.1596		
1120	(0.186)	(0.398)	(0.211)	0.555		
old38	0.008	0.072	-0.020	0.6662		
1.120	(0.184)	(0.365)	(0.214)	0.0002		
old39	-0.069	-0.074	-0.046	0.8993		
1140	(0.193)	(0.397)	(0.222)	0.0101		
old40	-0.353	0.160	-0.517	0.0101		
	(0.213)*	(0.379)	(0.263)*			

(continued on next page)

Table 6.3b (continued)

1141	0.001	0.107	0.100	0.0010
old41	-0.201	-0.186	-0.180	0.9812
1140	(0.210)	(0.411)	(0.246)	0.0014
old42	-0.321	0.315	-0.619	0.0014
1140	(0.226)	(0.381)	(0.292)*	0.6505
old43	-0.151	-0.064	-0.176	0.6507
	(0.215)	(0.442)	(0.248)	
old44	-0.478	-0.199	-0.557	0.2374
	(0.256)*	(0.489)	(0.303)*	
old45	-0.325	-0.177	-0.354	0.5407
	(0.252)	(0.520)	(0.290)	
old46	-0.332	0.004	-0.444	0.1512
	(0.262)	(0.490)	(0.312)	
old47	-0.839	-0.805	-0.823	0.9586
	(0.310)**	(0.757)	(0.343)**	
old48	-0.468	0.161	-0.590	0.0197
	(0.273)*	(0.523)	(0.322)*	
old49	-0.521	0.426	-0.973	0.0007
	(0.302)*	(0.467)	(0.414)**	
old50	-1.689	-1.244	-1.813	0.4338
	(0.595)**	(1.039)	(0.727)**	
old51	-1.343	-0.623	-1.597	0.1053
	(0.469)**	(0.763)	(0.601)**	
old52	-1.333	-1.464	-1.199	0.5859
	(0.440)**	(1.042)	(0.487)**	
old53	-1.823	-1.554	-1.919	0.6166
	(0.596)**	(1.042)	(0.729)**	
old54	-0.930	-0.758	-0.979	0.6195
	(0.383)**	(0.770)	(0.445)*	
old55	1.732	(01770)	2.138	_
	(0.187)**		(0.213)**	
old56	1 027	0.058	1 491	0.0000
01000	(0.222)**	(0.584)	(0.252)**	0.0000
old57	1 381	0.188	1 793	0.0000
01037	(0.205)**	(0.587)	(0.233)**	0.0000
01458	1.403	0.349	1.031	0.0000
01038	(0.200)**	(0.54)	(0.220)**	0.0000
01450	(0.209)	0.706	(0.239)	0.4636
01039	(0.702)	0.700	(0.200)**	0.4030
a1460	$(0.243)^{11}$	(0.400)	(0.288)**	0.0000
01000	3.313	3.987	3.099	0.0000
-1461	$(0.100)^{++}$	$(0.520)^{++}$	$(0.204)^{++}$	0.0000
01001	1.///	2.781	1.120	0.0000
-14(2)	$(0.245)^{**}$	(0.388)**	(0.300)**	0.0115
01062	2.039	2.0/1	1./91	0.0115
11/2	(0.261)**	(0.433)**	(0.348)**	0.0000
01d63	1.860	3.290	0.966	0.0000
	(0.318)**	(0.479)**	(0.486)*	
Old64	1.952	3.651	1.344	0.0000
	(0.243)**	(0.433)**	(0.309)**	
Observations	47027	24035	26374	

Table 6.3c

Education

	White-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
school_mis	0.118	0.311	0.050	0.1345	
	(0.149)	(0.309)	(0.175)		
school_1				-	
school_2	0.134	0.105	0.128	0.9215	
	(0.187)	(0.360)	(0.236)		
school_3	0.018	-0.079	-0.023	0.8224	
	(0.198)	(0.365)	(0.249)		
school_5	0.019	0.078	-0.011	0.7232	
	(0.215)	(0.432)	(0.251)		
school_6	-0.310	-0.503	-0.316	0.3176	
	(0.163)*	(0.352)	(0.187)*		
school_7	-0.142	-0.020	-0.208	0.3516	
	(0.169)	(0.339)	(0.202)		
school_8	-0.229	-0.171	-0.285	0.5217	
	(0.151)	(0.313)	(0.178)		
school_9	-0.095	-0.069	-0.152	0.6492	
	(0.155)	(0.319)	(0.183)		
Observations	47027	24035	26374		

Table 6.3d

Performance evaluation and career paths

	White-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
eval_mis	-0.300	-0.233		-	
	(0.391)	(0.408)			
eval_1	3.021	3.202	2.757	0.1127	
	(0.191)**	(0.286)**	(0.281)**		
eval_2	0.655	0.589	0.714	0.2953	
	(0.090)**	(0.141)**	(0.119)**		
eval_4	-0.075	-0.039	-0.086	0.5933	
	(0.072)	(0.125)	(0.089)		
eval_5	-0.633	-1.260	-0.698	0.2248	
	(0.343)*	(0.541)**	(0.463)		
eval_6				-	
Δ scale1	-0.385	-0.153	-0.351	0.1417	
	(0.116)**	(0.248)	(0.135)**		
Δ scale2	-0.395	-0.327	-0.372	0.7957	
	(0.147)**	(0.304)	(0.174)*		
Δ scale<0	-0.115	-39.435	-0.276	0.0000	
	(0.436)	(6.34e)	(0.445)		
d_scale<0	0.291	0.014	0.502	0.1577	
	(0.321)	(1.020)	(0.345)		
d_scale1	0.004	-0.143	-0.027	0.1932	
	(0.077)	(0.186)	(0.089)		
d_scale2	0.048	0.403	-0.131	0.0000	
	(0.091)	(0.232)*	(0.104)		
d_scale3	0.081	0.226	-0.073	0.0127	
	(0.103)	(0.257)	(0.120)		
promotion	-0.062	-0.536	-0.024	0.0000	
	(0.071)	(0.237)*	(0.075)		
prom_∆wage	0.017	0.019	0.021	0.8415	
	(0.010)*	(0.030)	(0.010)*		
degree	0.180	1.404	-0.171	0.0276	
	(0.543)	(0.894)	(0.715)		
degr_∆wage	0.025	0.032	0.016	0.8010	
	(0.047)	(0.079)	(0.063)		
Observations	47027	24035	26374		

Table 6.3e

Marital Status, ge	nder and work status
--------------------	----------------------

	White-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
Married	-0.261	-0.385	-0.218	0.0114	
	(0.051)**	(0.083)**	(0.066)**		
Divorced	-0.292	-0.364	-0.263	0.7618	
	(0.269)	(0.462)	(0.334)		
Male	-0.279	-0.250	-0.313	0.5323	
	(0.083)**	(0.155)	(0.100)**		
part_time	0.204	-0.128	0.290	0.0003	
	(0.103)*	(0.257)	(0.115)**		
Rehire	-0.226	-0.055	-0.377	0.1280	
	(0.183)	(0.409)	(0.212)*		
scale_12	-0.214	-0.049	-0.254	0.0381	
	(0.079)**	(0.147)	(0.099)**		
scale_14	0.035	0.143	0.032	0.3146	
	(0.088)	(0.153)	(0.110)		
scale_15	0.228	0.503	0.064	0.0004	
	(0.093)**	(0.154)**	(0.124)		
scale_16	0.203	0.460	0.024	0.0021	
	(0.105)*	(0.172)**	(0.142)		
scale_17	0.358	0.585	0.230	0.0455	
	(0.136)**	(0.226)**	(0.177)		
scale_18	0.158	0.710	-0.246	0.0000	
	(0.140)	(0.220)**	(0.190)		
scale_19	-0.061	0.697	-0.505	0.0002	
	(0.233)	(0.350)*	(0.321)		
Observations	47027	24035	26374		

Table 6.3f

Location

	White-Collar Workers				
	1987-1996	1987-1991	1991-1996	p-value	
location1	0.779	0.668		-	
	(0.324)**	(0.335)*			
location2	0.429	0.332	0.459	0.2671	
	(0.092)**	(0.163)*	(0.115)**		
location3	-0.486	-0.088	-0.785	0.0000	
	(0.095)**	(0.141)	(0.131)**		
location4	0.043	0.295	-0.014	0.1308	
	(0.161)	(0.271)	(0.205)		
location5	0.319	0.459	0.216	0.0859	
	(0.114)**	(0.197)**	(0.142)		
location6	0.549	0.669	0.516	0.1862	
	(0.095)**	(0.173)**	(0.116)**		
location7	-0.008	-0.059	-0.031	0.8394	
	(0.117)	(0.243)	(0.138)		
Observations	47027	24035	26374		

Number of workers of workers employed at the 14th of each month during the period from January 1987 until April 1996













Panel d) Distribution of schooling degrees among workers separating in the period January 1st 1987-March 1st 1991



Panel e) Distribution of performance evaluation scores among employed workers on March 1st 1991



Panel f) Distribution of employed workers in wage scales on March 1st 1991



Panel a) Distribution of locations from which workers separate in the period January 1st 1987 - March 1st 1991









Panel d) Distribution of schooling degrees among workers separating in the period January 1st 1987-March 1st 1991







Panel f) Distribution of wage scales of workers separating in the period January 1st 1987-March 1st 1991



Panel a) Distribution of locations from which workers separate in the period March 1st 1991 – March 14th 1996









Panel d) Distribution of schooling degrees among workers separating in the period March 1st 1991 – March 14th 1996



Panel e) Distribution of performance evaluation scores of workers separating in the period March 1st 1991 – March 14th 1996



Panel f) Distribution of wage scales of workers separating in the period March 1st 1991 – March 14th 1996




Panel b) Baseline Hazards estimated separately over the periods 1987-1991 and 1991-1996 for the entire sample (corresponding parameter estimates are in Table 6.1, column 2 and 3)



69

Panel a) Baseline hazard for **blue-collar workers** estimated over the period 1987-1996 when baseline hazards are allowed to differ for blue- and white-collar workers while coefficients are restricted to be the same.







.01

.005

.001

Ò

365



Panel b) Baseline hazard for **blue-collar workers** only estimated separately over the periods 1987-1991 and 1991-1996 (corresponding parameter estimates are in Table 6.2, columns 2 and 3).

പെക്കാപ്പും

2557

2922

3287

6 1461 1826 2192 Day of Separation

<u>୶୶୳୶୳୶</u>୲୶୶୶୶୶୶୶୶୶

1096

731





Panel b) Baseline hazard for **white-collar workers** only estimated separately over the periods 1987-1991 and 1991-1996 (corresponding parameter estimates are in Table 6.3, columns 2 and 3).





Figure 6.5a Age Effects in Different Periods (Coefficient Estimated from Table 6.1)

Age in Years



Figure 6.5b Age Effects for Blue-Collar Workers in Different Periods (Coefficient Estimates from Table 6.2)

Age in Years



Figure 6.5c Age Effects for White-Collar Workers in Different Periods

Age in Years

Appendix

Definition of Variables

Location/Subsidiary

We define eight dummy variables that equal 1 if the worker is employed at the respective location and equal 0 else:

location0	Fokker Aircraft (Schiphol) or Fokker Administration
	(Amsterdam)
location1	Aerostructures (Schiphol)
location2	Fokker Aircraft (Papendrecht/Dordrecht)
location3	Fokker Space and Systems (Leiden)
location4	Fokker ELMO (Woensdrecht)
location5	Fokker Special Products (Hoogeveen)
location6	Fokker Aircraft (Ypenburg)
location7	Fokker Aircraft Services (Woensdrecht)

Education

The categorical dummy variables we define equal 1 for the highest educational or vocational degree that a worker has completed at the time of observation and 0 else.

school_1	basic education (lo)
school_2	lower vocational schooling degree (lbo)
school_3	intermediate general schooling degree (mbo)
school_4	apprenticeship. (leerlingwezen)
school_5	higher general schooling degree (havo)
school_6	intermediate vocational schooling degree (mbo)
school_7	general schooling degree that qualifies to attend university
(vwo)	
school_8	higher vocational schooling degree (hbo)
school_9	university
school_mis	information on educational and vocational achievement is
	missing.

Note that schooling degrees, such as basic and intermediate, higher general schooling degrees or the degree that entitles to attend university (categories 1,3,5 and 7), are prerequisites for pursuing a given vocational or general education in the Dutch educational system. Basic education is a prerequisite for any other degree and one can decide after having completed basic education to either follow a lower vocational schooling course or to attend any of the school forms leading to any other general schooling degree. Intermediate education makes one eligible to follow intermediate

vocational training or complete an apprenticeship. An intermediate general schooling degree qualifies for higher vocational schooling, a higher general schooling degree (havo) qualifies for higher vocational schooling (hbo), while the highest level general schooling degree is a prerequisite for pursuing a college or university degree. In addition, it is possible to pursue the next higher schooling level after having obtained a given schooling degree; similarly it is possible to enter the next higher level of vocational schooling after having completed vocational schooling at the level just below, e.g. after having completed intermediate vocational schooling one is eligible to enter higher vocational schooling.

Age

We define dummy variables that equal 1 if the condition is true and 0 else.

old 17	if younger than 18 years of age
old'i' for 18≤'i'≤63	if 'i' years of age
old64	if 64 years of age or older

Tenure

We define 16 categorical dummy variables that equal 1 if the respective condition is true and zero else.

ten3m	if tenure 0-3 months
ten6m	if tenure 3-6 months
ten9m	if tenure 6-9 months
ten12m	if tenure 9-12 months
ten1	if tenure 1-2 years
ten2	if tenure 2-3 years
ten3	if tenure 3-4 years
ten4	if tenure 4-5 years
ten5_9	if tenure 5-10 years
ten10_14	if tenure 10-15years
ten15_19	if tenure 15-20 years
ten20_24	if tenure 20-25 years
ten25_29	if tenure 25-30 years
ten30_34	if tenure 30-35 years
ten35_39	if tenure 35-40 years
ten40	if tenure more than 40 years

Performance Evaluation

The dummy variables equal 1 if the worker received the respective performance evaluation score, and equal 0 else.

eval_1	unsatisfactory
eval_2	satisfactory
eval_3	good
eval_4	very good
eval_5	outstanding
eval_6	excellent
eval_mis	evaluation score is missing

Wage Scales and Transitions in Wage Scales

scale_'i' for 2≤i≤19	if 'i' = the first observed wage scale (i.e. the wage scale on 1st January 1987 if employed then or the wage scale at hiring if hired after 1st of January 1987)
∆scale1	1 if the worker moves up one wage scale (the dummy equals 1 only for the observation at which the change occurs)
∆scale2	1 if the worker moves up more than one wage scale (the dummy equals 1 only for the observation at which the change occurs)
Δ scale<0	1 if the worker moves down at least one wage scale (the dummy equals 1 only for the observation at which the change occurs)
d_scale1	1 if the worker has moved up one wage scale since the first recording for that worker
d_scale2	1 if the worker has moved up two wage scales since the first recording for that worker
d_scale3	1 if the worker has moved up more than two wage scales since the first recording for that worker
d_scale<0	1 if the worker has moved up down at least one wage scale since the first recording for that worker

Promotions, Degree Completion and Additional Wage Growth

promotion	counts the number of contract changes because of a
	promotion from 1st January 1987 onwards.
prom_∆wage	adds up percentage points of additional wage growth that
	was associated with a promotion
degree	counts the number of contract changes because of a degree
	completion from 1st January 1987 onwards
degr_∆wage	adds up percentage points of additional wage growth that
	was associated with a degree completion

Gender, Marital Status, Worker Status

married	1 if married, 0 if divorced or else
divorced	1 if divorced, 0 else
male	1 if male, 0 else
part_time	1 if the employee works part-time, 0 else
rehire	1 if the worker had previously been employed at the firm
	and is rehired during the observation period, 0 else

IZA Discussion Papers

No.	Author(s)	Title	Area	Date
101	L. Husted H. S. Nielsen M. Rosholm N. Smith	Employment and Wage Assimilation of Male First Generation Immigrants in Denmark	3	1/00
102	B. van der Klaauw J. C. van Ours	Labor Supply and Matching Rates for Welfare Recipients: An Analysis Using Neighborhood Characteristics	2/3	1/00
103	K. Brännäs	Estimation in a Duration Model for Evaluating Educational Programs	6	1/00
104	S. Kohns	Different Skill Levels and Firing Costs in a Matching Model with Uncertainty – An Extension of Mortensen and Pissarides (1994)	1	1/00
105	G. Brunello C. Graziano B. Parigi	Ownership or Performance: What Determines Board of Directors' Turnover in Italy?	1	1/00
106	L. Bellmann S. Bender U. Hornsteiner	Job Tenure of Two Cohorts of Young German Men 1979 - 1990: An analysis of the (West-)German Employment Statistic Register Sample concerning multivariate failure times and unobserved heterogeneity	1	1/00
107	J. C. van Ours G. Ridder	Fast Track or Failure: A Study of the Completion Rates of Graduate Students in Economics	5	1/00
108	J. Boone J. C. van Ours	Modeling Financial Incentives to Get Unemployed Back to Work	3/6	1/00
109	G. J. van den Berg B. van der Klaauw	Combining Micro and Macro Unemployment Duration Data	3	1/00
110	D. DeVoretz C. Werner	A Theory of Social Forces and Immigrant Second Language Acquisition	1	2/00
111	V. Sorm K. Terrell	Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic	1/4	2/00
112	L. Bellmann T. Schank	Innovations, Wages and Demand for Heterogeneous Labour: New Evidence from a Matched Employer-Employee Data-Set	5	2/00
113	R. Euwals	Do Mandatory Pensions Decrease Household Savings? Evidence for the Netherlands	3	2/00
114	G. Brunello A. Medio	An Explanation of International Differences in Education and Workplace Training	2	2/00
115	A. Cigno F. C. Rosati	Why do Indian Children Work, and is it Bad for Them?	3	2/00

116	C. Belzil	Unemployment Insurance and Subsequent Job Duration: Job Matching vs. Unobserved Heterogeneity	3	2/00
117	S. Bender A. Haas C. Klose	IAB Employment Subsample 1975-1995. Opportunities for Analysis Provided by the Anonymised Subsample	7	2/00
118	M. A. Shields M. E. Ward	Improving Nurse Retention in the British National Health Service: The Impact of Job Satisfaction on Intentions to Quit	5	2/00
119	A. Lindbeck D. J. Snower	The Division of Labor and the Market for Organizations	5	2/00
120	P. T. Pereira P. S. Martins	Does Education Reduce Wage Inequality? Quantile Regressions Evidence from Fifteen European Countries	5	2/00
121	J. C. van Ours	Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?	4/6	3/00
122	D. Munich J. Svejnar K. Terrell	Returns to Human Capital under the Communist Wage Grid and During the Transition to a Market Economy	4	3/00
123	J. Hunt	Why Do People Still Live in East Germany?	1	3/00
124	R. T. Riphahn	Rational Poverty or Poor Rationality? The Take-up of Social Assistance Benefits	3	3/00
125	F. Büchel J. R. Frick	The Income Portfolio of Immigrants in Germany - Effects of Ethnic Origin and Assimilation. Or: Who Gains from Income Re-Distribution?	1/3	3/00
126	J. Fersterer R. Winter-Ebmer	Smoking, Discount Rates, and Returns to Education	5	3/00
127	M. Karanassou D. J. Snower	Characteristics of Unemployment Dynamics: The Chain Reaction Approach	3	3/00
128	O. Ashenfelter D. Ashmore O. Deschênes	Do Unemployment Insurance Recipients Actively Seek Work? Evidence From Randomized Trials in Four U.S. States	6	3/00
129	B. R. Chiswick M. E. Hurst	The Employment, Unemployment and Unemployment Compensation Benefits of Immigrants	1/3	3/00
130	G. Brunello S. Comi C. Lucifora	The Returns to Education in Italy: A New Look at the Evidence	5	3/00
131	B. R. Chiswick	Are Immigrants Favorably Self-Selected? An Economic Analysis	1	3/00
132	R. A. Hart	Hours and Wages in the Depression: British Engineering, 1926-1938	7	3/00
133	D. N. F. Bell R. A. Hart O. Hübler W. Schwerdt	Paid and Unpaid Overtime Working in Germany and the UK	1	3/00

134	A. D. Kugler G. Saint-Paul	Hiring and Firing Costs, Adverse Selection and Long-term Unemployment	3	3/00
135	A. Barrett P. J. O'Connell	Is There a Wage Premium for Returning Irish Migrants?	1	3/00
136	M. Bräuninger M. Pannenberg	Unemployment and Productivity Growth: An Empirical Analysis within the Augmented Solow Model	3	3/00
137	JSt. Pischke	Continuous Training in Germany	5	3/00
138	J. Zweimüller R. Winter-Ebmer	Firm-specific Training: Consequences for Job Mobility	1	3/00
139	R. A. Hart Y. Ma	Wages, Hours and Human Capital over the Life Cycle	1	3/00
140	G. Brunello S. Comi	Education and Earnings Growth: Evidence from 11 European Countries	2/5	4/00
141	R. Hujer M. Wellner	The Effects of Public Sector Sponsored Training on Individual Employment Performance in East Germany	6	4/00
142	J. J. Dolado F. Felgueroso J. F. Jimeno	Explaining Youth Labor Market Problems in Spain: Crowding-Out, Institutions, or Technology Shifts?	3	4/00
143	P. J. Luke M. E. Schaffer	Wage Determination in Russia: An Econometric Investigation	4	4/00
144	G. Saint-Paul	Flexibility vs. Rigidity: Does Spain have the worst of both Worlds?	1	4/00
145	MS. Yun	Decomposition Analysis for a Binary Choice Model	7	4/00
146	T. K. Bauer J. P. Haisken-DeNew	Employer Learning and the Returns to Schooling	5	4/00
147	M. Belot J. C. van Ours	Does the Recent Success of Some OECD Countries in Lowering their Unemployment Rates Lie in the Clever Design of their Labour Market Reforms?	3	4/00
148	L. Goerke	Employment Effects of Labour Taxation in an Efficiency Wage Model with Alternative Budget Constraints and Time Horizons	3	5/00
149	R. Lalive J. C. van Ours J. Zweimüller	The Impact of Active Labor Market Programs and Benefit Entitlement Rules on the Duration of Unemployment	3/6	5/00
150	J. DiNardo K. F. Hallock JSt. Pischke	Unions and the Labor Market for Managers	7	5/00
151	M. Ward	Gender, Salary and Promotion in the Academic Profession	5	5/00

152	J. J. Dolado F. Felgueroso J. F. Jimeno	The Role of the Minimum Wage in the Welfare State: An Appraisal	3	5/00
153	A. S. Kalwij M. Gregory	Overtime Hours in Great Britain over the Period 1975-1999: A Panel Data Analysis	3	5/00
154	M. Gerfin M. Lechner	Microeconometric Evaluation of the Active Labour Market Policy in Switzerland	6	5/00
155	J. Hansen	The Duration of Immigrants' Unemployment Spells: Evidence from Sweden	1/3	5/00
156	C. Dustmann F. Fabbri	Language Proficiency and Labour Market Per- formance of Immigrants in the UK	1	5/00
157	P. Apps R. Rees	Household Production, Full Consumption and the Costs of Children	7	5/00
158	A. Björklund T. Eriksson M. Jäntti O. Raaum E. Österbacka	Brother Correlations in Earnings in Denmark, Finland, Norway and Sweden Compared to the United States	5	5/00
159	P J. Jost M. Kräkel	Preemptive Behavior in Sequential Tournaments	5	5/00
160	M. Lofstrom	A Comparison of the Human Capital and Signaling Models: The Case of the Self-Employed and the Increase in the Schooling Premium in the 1980's	5	6/00
161	V. Gimpelson D. Treisman G. Monusova	Public Employment and Redistributive Politics: Evidence from Russia's Regions	4	6/00
162	C. Dustmann M. E. Rochina- Barrachina	Selection Correction in Panel Data Models: An Application to Labour Supply and Wages	6	6/00
163	R. A. Hart Y. Ma	Why do Firms Pay an Overtime Premium?	5	6/00
164	M. A. Shields S. Wheatley Price	Racial Harassment, Job Satisfaction and Intentions to Quit: Evidence from the British Nursing Profession	5	6/00
165	P. J. Pedersen	Immigration in a High Unemployment Economy: The Recent Danish Experience	1	6/00
166	Z. MacDonald M. A. Shields	The Impact of Alcohol Consumption on Occupa- tional Attainment in England	5	6/00
167	A. Barrett J. FitzGerald B. Nolan	Earnings Inequality, Returns to Education and Immigration into Ireland	5	6/00
168	G. S. Epstein A. L. Hillman	Social Harmony at the Boundaries of the Welfare State: Immigrants and Social Transfers	3	6/00

169	R. Winkelmann	Immigration Policies and their Impact: The Case of New Zealand and Australia	1	7/00
170	T. K. Bauer K. F. Zimmermann	Immigration Policy in Integrated National Economies	1	7/00
171	C. Dustmann F. Windmeijer	Wages and the Demand for Health – A Life Cycle Analysis	5	7/00
172	D. Card	Reforming the Financial Incentives of the Welfare System	3	7/00
173	D. S. Hamermesh	Timing, Togetherness and Time Windfalls	5	7/00
174	E. Fehr JR. Tyran	Does Money Illusion Matter? An Experimental Approach	7	7/00
175	M. Lofstrom	Self-Employment and Earnings among High- Skilled Immigrants in the United States	1	7/00
176	O. Hübler W. Meyer	Industrial Relations and the Wage Differentials between Skilled and Unskilled Blue-Collar Workers within Establishments: An Empirical Analysis with Data of Manufacturing Firms	5	7/00
177	B. R. Chiswick G. Repetto	Immigrant Adjustment in Israel: Literacy and Fluency in Hebrew and Earnings	1	7/00
178	R. Euwals M. Ward	The Renumeration of British Academics	5	7/00
179	E. Wasmer P. Weil	The Macroeconomics of Labor and Credit Market Imperfections	2	8/00
180	T. K. Bauer I. N. Gang	Sibling Rivalry in Educational Attainment: The German Case	5	8/00
181	E. Wasmer Y. Zenou	Space, Search and Efficiency	2	8/00
182	M. Fertig C. M. Schmidt	Discretionary Measures of Active Labor Market Policy: The German Employment Promotion Reform in Perspective	6	8/00
183	M. Fertig C. M. Schmidt	Aggregate-Level Migration Studies as a Tool for Forecasting Future Migration Streams	1	8/00
184	M. Corak B. Gustafsson T. Österberg	Intergenerational Influences on the Receipt of Unemployment Insurance in Canada and Sweden	3	8/00
185	H. Bonin K. F. Zimmermann	The Post-Unification German Labor Market	4	8/00
186	C. Dustmann	Temporary Migration and Economic Assimilation	1	8/00

187	T. K. Bauer M. Lofstrom K. F. Zimmermann	Immigration Policy, Assimilation of Immigrants and Natives' Sentiments towards Immigrants: Evidence from 12 OECD-Countries	1	8/00
188	A. Kapteyn A. S. Kalwij A. Zaidi	The Myth of Worksharing	5	8/00
189	W. Arulampalam	Is Unemployment Really Scarring? Effects of Unemployment Experiences on Wages	3	8/00
190	C. Dustmann I. Preston	Racial and Economic Factors in Attitudes to Immigration	1	8/00
191	G. C. Giannelli C. Monfardini	Joint Decisions on Household Membership and Human Capital Accumulation of Youths: The role of expected earnings and local markets	5	8/00
192	G. Brunello	Absolute Risk Aversion and the Returns to Education	5	8/00
193	A. Kunze	The Determination of Wages and the Gender Wage Gap: A Survey	5	8/00
194	A. Newell F. Pastore	Regional Unemployment and Industrial Restructuring in Poland	4	8/00
195	F. Büchel A. Mertens	Overeducation, Undereducation, and the Theory of Career Mobility	5	9/00
196	J. S. Earle K. Z. Sabirianova	Equilibrium Wage Arrears: A Theoretical and Empirical Analysis of Institutional Lock-In	4	9/00
197	G. A. Pfann	Options to Quit	1	9/00
198	M. Kreyenfeld C. K. Spiess G. G. Wagner	A Forgotten Issue: Distributional Effects of Day Care Subsidies in Germany	3	9/00
199	H. Entorf	Rational Migration Policy Should Tolerate Non- Zero Illegal Migration Flows: Lessons from Modelling the Market for Illegal Migration	1	9/00
200	T. Bauer G. S. Epstein I. N. Gang	What are Migration Networks?	1	9/00
201	T. J. Dohmen G. A. Pfann	Worker Separations in a Nonstationary Corporate Environment	1	9/00